

Efficient Modeling of Gravity Fields Caused by Sources with Arbitrary Geometry and Arbitrary Density Distribution

Leyuan Wu¹

Received: 25 August 2017/Accepted: 8 January 2018/Published online: 20 January 2018 © Springer Science+Business Media B.V., part of Springer Nature 2018

Abstract We present a brief review of gravity forward algorithms in Cartesian coordinate system, including both space-domain and Fourier-domain approaches, after which we introduce a truly general and efficient algorithm, namely the convolution-type Gauss fast Fourier transform (Conv-Gauss-FFT) algorithm, for 2D and 3D modeling of gravity potential and its derivatives due to sources with arbitrary geometry and arbitrary density distribution which are defined either by discrete or by continuous functions. The Conv-Gauss-FFT algorithm is based on the combined use of a hybrid rectangle-Gaussian grid and the fast Fourier transform (FFT) algorithm. Since the gravity forward problem in Cartesian coordinate system can be expressed as continuous convolution-type integrals, we first approximate the continuous convolution by a weighted sum of a series of shifted discrete convolutions, and then each shifted discrete convolution, which is essentially a Toeplitz system, is calculated efficiently and accurately by combining circulant embedding with the FFT algorithm. Synthetic and real model tests show that the Conv-Gauss-FFT algorithm can obtain high-precision forward results very efficiently for almost any practical model, and it works especially well for complex 3D models when gravity fields on large 3D regular grids are needed.

Keywords Convolution-type integrals · Gravity forward modeling · Conv-Gauss-FFT algorithm · Arbitrary geometry · Arbitrary density distribution

Leyuan Wu leyuanwu@zjut.edu.cn

¹ Center for Optics and Optoelectronics Research (COOR), Collaborative Innovation Center for Information Technology in Biological and Medical Physics, College of Science, Zhejiang University of Technology, Hangzhou 310023, China

1 Introduction

The gravity forward problem is the foundation of gravity exploration in geophysics. It can be formulated either in spherical coordinate system for global- and regional-scale problems, or in Cartesian coordinate system for local problems when the planar approximation of the earth surface is acceptable. In the former case, the forward problem can be solved either in the spectral domain using fast algorithms for spherical harmonic expansions (Driscoll and Healy 1994; Mohlenkamp 1999; Rokhlin and Tygert 2005; Tygert 2006; Gruber et al. 2011), or in the space domain using solutions of a spherical prism, i.e., a tesseroid (Asgharzadeh et al. 2007; Heck and Seitz 2007; Grombein et al. 2013; Uieda et al. 2016; Deng et al. 2016). In the latter case, analogously, the forward problem is usually solved in the Fourier domain based on the FFT algorithm (Parker 1973; Pedersen 1978; Forsberg 1985; Hansen and Wang 1988; Sideris and Li 1993; Li and Sideris 1994; Wu and Tian 2014), or in the space domain using solutions of a rectangular prism (Nagy 1966; Li and Chouteau 1998; Nagy et al. 2000).

Many authors have made contributions to the gravity forward problem, presenting various approaches involving both analytical, numerical and hybrid analytical–numerical (seminumerical) algorithms. In our opinion, these contributions may be assigned to four main aspects:

- 1. to model sources with more arbitrary geometries;
- 2. to model sources with more general density distributions;
- to model the gravity potential and its first-order (the gravity vector), second-order (the gravity gradient tensor) and higher-order derivative components;
- 4. to model the gravity effects listed above in all space, both outside and within the source body, and to clarify singularity problems.

Obviously, it is a universal solution which can be used for sources with arbitrary complicated geometry and arbitrary variable density distribution that we are pursuing. Moreover, we always prefer algorithms with higher accuracy and lower computational costs, though most of the time we have to find a reasonable trade-off between these two factors. In the following, we provide a brief review of existing forward algorithms formulated in the Cartesian coordinate system, as the method we present in this paper generally falls into this category. However, it should be mentioned that some models, e.g., the polyhedron, which is originally formulated in the Cartesian coordinate system through a coordinate system transformation.

Historically, space-domain solutions have played a major role in the gravity modeling theory. Both analytical and numerical techniques are used whenever they are well suited for the model under consideration. Analytical solutions are very useful; they are considered as the most accurate algorithm and they can provide references to numerical or seminumerical algorithms. However, they are available only when the source body is assumed to have relatively simple geometry and density distribution. When some complicated models are to be calculated, e.g., sources with their boundaries or density distributions described by some arbitrary functions, numerical or seminumerical solutions become necessary.

For the general 2D gravity forward problem, a source body with its cross section represented by an arbitrary polygon is usually considered as the general case of source geometry. Analytical expressions of the gravity anomaly due to 2D polygonal sources with homogeneous density were derived by Talwani et al. (1959). Later, analytical solutions of variable density models, including linear models (Murthy and Rao 1979), quadratic

polynomial models (Rao 1986), hyperbolic models (Rao et al. 1994) and general 2D polynomial models (Zhang et al. 2001; Zhou 2010), were also derived. Jia and Wu (2011) revised forward modeling of the gravity-field components and their first- and second-order partial derivatives due to homogeneous 2D polygonal sources and addressed some singularity cases. D'Urso (2015) provided a nice review of gravity forward algorithms due to a 2D polygonal source whose density contrast is a 2D polynomial function.

In addition to analytical solutions, numerical solutions are also provided for polygonal sources with variable density. Numerical techniques become indispensable when a certain density function, like the exponential function (Cordell 1973), or some other more arbitrary functions are used (Zhou 2008, 2009b). They also provide a simple solution when sources with continuous boundaries are studied (Martin-Atienza and Garcia-Abdeslem 1999; Garcia-Abdeslem 2003). Although analytical solutions are not available under these conditions, it should be mentioned that the polygonal model with 2D polynomial density function has in fact provided a quite general solution for 2D models, because any continuous boundary can be well approximated by a n-sided polygon when n is sufficiently large, and any 2D density distribution can be well approximated by a 2D polynomial density function when the order of the polynomial is sufficiently high.

For the general 3D gravity forward problem, previous works mainly focus on two models: the right rectangular prism and the polyhedron. A combination of the two, namely the rectangular prism with inclined top and bottom faces, has also been studied by several authors (Smith 2000; Tsoulis et al. 2003; D'Urso and Trotta 2015).

The right rectangular prism is probably the most widely used model in gravity exploration. A collection of prisms provides a simple way to approximate a source body with complex geometry and arbitrary variable density contrast. Nagy et al. (2000) presented closed-form expressions of the gravity potential, and its derivatives, up to the third order for a prism, in a unified manner, and provided a detailed singularity analysis of these expressions when they are generalized to the whole space. Later, analytical solutions due to the rectangular prism with several depth-dependent variable density contrast, including the parabolic model (Chakravarthi et al. 2002), the quadratic polynomial model (Gallardo-Delgado et al. 2003; Gallardo et al. 2005) and the cubic polynomial model (Garcia-Abdeslem 2005), were also studied. Recently, Jiang et al. (2017) derived analytical solution of gravity anomalies due to a right rectangular prism with an arbitrary order of polynomial density-contrast function of depth. Soon after, Zhang and Jiang (2017) further extended the solution to a linear combination of three arbitrary-order polynomial density functions in x-, y- and z-directions. Although these models can already cover a wide range of density distributions, when more complex models are considered, such as depth-dependent density models described by arbitrary functions, or density models vary both horizontally and vertically following some arbitrary functions, numerical or hybrid solutions are still useful (Garcia-Abdeslem 1992; Zhou 2009a).

Comparing to the right rectangular prism, the polyhedron offers a more general geometric representation at the expense of a more complicated derivation for the analytical solutions. Closed-form solutions for the gravity anomaly due to homogeneous polyhedral bodies have been studied by many authors (Barnett 1976; Okabe 1979; Holstein and Ketteridge 1996; Guptasarma and Singh 1999; Tsoulis and Petrovic 2001; Tsoulis 2012; D'Urso 2013, 2014a; Conway 2015; Werner 2017). Polyhedral bodies with linear density distribution have also received considerable attention during the past decade (Pohanka 1998; Hansen 1999; Holstein 2003; Hamayun and Tenzer 2009; D'Urso 2014b, 2016). Ren et al. (2017a) derived singularity-free analytic formula of the gravity potential caused by polyhedral bodies with its density varies in the case of $\lambda = ax^m + by^n + cz^t$, $m \le 2, n \le 2, t \le 2$, and analytic formula of the gravity potential due to a rectangular prism with $m \le 3, n \le 3, t \le 3$. Recently, D'Urso and Trotta (2017) and Ren et al. (2017c) presented singularity-free expressions for evaluating the gravity anomaly due to polyhedral bodies with its density contrast described by a third-order polynomial, and indicated that the general approach can be easily extended to higher-order polynomial functions. Analogous to the 2D case, such a model may provide a quite general solution for 3D forward problems since a polyhedron with increasing number of facets can be used to approximate a 3D source with arbitrary geometry, and a 3D polynomial density function with increasing order can be used to approximate as closely as desired any smooth density distribution.

In addition to space-domain methods, Fourier-domain methods have also been developed. Compared to analytical solutions, Fourier-domain solutions are considered to be less accurate. However, they are far more computationally efficient, thanks to the fast Fourier transform algorithm (Cooley and Tukey 1965), especially when gravity fields on large regular grids are to be calculated. Besides, Fourier-domain expressions of the gravity potential and its derivatives are related through very simple multiplicative factors. Therefore, computational costs can be further reduced for Fourier-domain algorithms when multiple gravity fields, including vector and tensor components, or even higher-order derivatives, are calculated simultaneously (Wu and Chen 2016).

Fourier-domain solutions are available for a single rectangular prism with homogeneous density distribution (Bhattacharyya 1966), depth-dependent linear or exponential density distributions (Chai and Hinze 1988; Chenot and Debeglia 1990; Lee and Biehler 1991), and general 3D polynomial density distributions (Wu and Chen 2016). They are also available for 2D prismatic bodies with general 2D polynomial density distributions (Wu and Chen 2016), and arbitrary 2D polygonal bodies with homogeneous density distribution (Pedersen 1978). Fourier-domain solutions due to arbitrary 3D polyhedral bodies in the case of homogeneous density contrast were derived by Wu (1981) and Pedersen (1978), and later simplified by Hansen and Wang (1988). For variable polyhedral models, Wu (1983) derived Fourier-domain expressions of the gravity anomaly caused by an arbitrary polyhedron with linear and exponential depth-dependent density function.

In addition to single-source bodies, Fourier-domain algorithms work especially efficiently for large-scale combined models, e.g., a 2D layer with horizontally variable density distribution (Blakely 1996), a general 3D source with arbitrarily variable density distribution (Tontini et al. 2009; Sanso and Sideris 2013; Zhang and Wong 2015) and topographic models with density variations both horizontally and vertically, following exponential, parabolic or polynomial functions (Parker 1973; Forsberg 1985; Granser 1987; Chai and Hinze 1988; Chai and Jia 1990; Guspi 1992; Sideris and Li 1993; Li and Sideris 1994; Chappell and Kusznir 2008; Zhang et al. 2015; Wu 2016). These models are frequently used in forward problems such as gravity terrain correction, isostatic compensation and inverse problems such as equivalent source construction, 3D density imaging, and sediment–basement interface or crust–mantle interface determinations. Wu (2016) provided a brief review of Fourier-domain algorithms for the modeling of topographic masses, in which they were divided into three categories, namely Chai-type algorithms (Chai and Hinze 1988), Parker-type algorithms (Parker 1973) and Forsbergtype algorithms (Forsberg 1985).

Alternatively, the gravity forward problem can also be solved by finite differences method (Farquharson and Mosher 2009), finite element methods (Cai and Wang 2005), Cauchy-type integrals (Zhdanov and Liu 2013; Cai and Zhdanov 2015) and fast multipole method (FMM) (Casenave et al. 2016; Ren et al. 2017b).

In this paper, we focus on solving the general 2D and 3D gravity forward problems in Cartesian coordinate system using Fourier-domain algorithms. It is well known that the general gravity forward problem can be expressed as continuous convolution-type integrals (Bhattacharyya and Navolio 1975; Blakely 1996). The gravity potential and its derivatives, including the gravity vector, the gravity gradient tensor, and higher-order components like the gravitational curvatures, can all be written as the convolution of the density distribution function and the corresponding kernel function, that is, Newton's integral kernel, or its derivative of certain order.

To evaluate the continuous convolution integral using the discrete FFT algorithm, there are two different approaches; both involve two major steps. The first approach begins by transforming the continuous convolution in the space domain to the equivalent point-wise multiplication in the Fourier domain based on the convolution theorem. Then both the forward and inverse continuous Fourier transforms are calculated approximately using the forward and inverse discrete FFT algorithms, respectively. As the continuous Fourier transform of Newton's integral kernel or its derivatives can be obtained analytically, the forward transform is applied only to the density function, while the inverse transform is applied to the Fourier-domain product of the density and kernel functions. Since the discrete FFT algorithm is in fact the rectangle rule of the continuous Fourier transform after truncating at finite integral limits (Wu and Tian 2014), the forward results contain both truncation and quadrature errors (Wu and Chen 2016). The second approach begins by approximating the continuous convolution integral using a uniformly sampled discrete convolution, which is also the rectangle rule. Then the discrete convolution, which is intrinsically a Toeplitz matrix-vector product, is embedded into a circulant matrix-vector product and evaluated efficiently and accurately, to almost machine precision, through the FFT algorithm (Vogel 2002). Likewise, the approximation of the continuous convolution using a discrete one will inevitably produce quadrature errors.

Fourier-domain algorithms including Parker-type algorithms (Parker 1973; Wu 2016) for topographic models, and the 3D FFT-based algorithm introduced in Tontini et al. (2009) for general 3D models, generally belong to the first approach; while Forsberg-type algorithms (Forsberg 1985) and those introduced in Sanso and Sideris (2013) and Zhang and Wong (2015) generally belong to the second approach. It should be mentioned that when a prism-stacked model is used, as is the case in Sanso and Sideris (2013), the continuous convolution integral reduces to a discrete convolution after integrating within each grid cell and replacing the point mass kernel function with the rectangular prism kernel function. Then the results obtained by the second approach mentioned above agree with the classical prism summation results to almost machine precision. However, when continuous density models are considered, that is, the density varies not only from prism to prism, but also within each prism element, both approaches involve a discretization step where the rectangle rule is applied to approximate a continuous integral, i.e., the continuous convolution integral or the continuous Fourier transform integral, and this discretization step produces numerical errors, which becomes one of the greatest drawbacks of Fourier-domain solutions.

To improve the numerical accuracy of Fourier-domain approaches, numerical quadrature errors need to be reduced. For the first approach mentioned above, we have already introduced a Gauss-FFT algorithm to calculate more accurately the forward and inverse continuous Fourier transforms (Wu and Tian 2014; Wu and Chen 2016; Wu 2016; Wu and Lin 2017). The Gauss-FFT algorithm is based on a hybrid rectangle-Gaussian sampling of the function to be transformed. It converges to the precise results much faster than the standard FFT algorithm with grid expansion, which is in fact the rectangle rule with smaller steps (Wu and Tian 2014). Here in this paper, we try to focus on the second Fourier-domain approach. Inspired by the Gauss-FFT algorithm, a Conv-Gauss-FFT algorithm is introduced by applying the hybrid rectangle-Gaussian grid directly to the continuous convolution integral. Instead of using a classical rectangle rule, we first approximate the continuous convolution integral using a weighted sum of a series of shifted rectangle rules, with the weights and offsets determined by a Gaussian quadrature rule. Then each shifted rectangle rule, which is basically a discrete convolution, is evaluated efficiently and accurately, to almost machine precision, through the FFT algorithm (Vogel 2002).

The Conv-Gauss-FFT algorithm is rigorously the hybrid rectangle-Gaussian quadrature rule for the continuous convolution integral, with its numerical accuracy controlled by both the step length of the rectangle grid and number of Gaussian nodes within each rectangle interval. Together with a point-in-polygon or a point-in-polyhedron algorithm, it can be easily applied to solve general 2D and 3D gravity forward problems involving sources with arbitrary geometry and arbitrary variable density distribution. Generally speaking, the algorithm can achieve high accuracy for field points considerably far from the source, but it may introduce some observable errors for field points close to the source's boundary.

This paper is organized as follows: Section 2 introduces a hybrid rectangle-Gaussian grid for continuous convolutions. The 1D, 2D and 3D continuous convolutions are calculated using a weighted sum of a series of discrete convolutions. Section 3 provides some details on how to evaluate discrete convolutions efficiently and accurately by combining circulant embedding with the FFT algorithm. Section 4 presents expressions of the 1D, 2D and 3D Conv-Gauss-FFT algorithms based on the results obtained in Sects. 2 and 3 and provides some general discussions on the numerical accuracy and computational complexity of the presented algorithm. In Sect. 5, the 2D and 3D gravity forward problems are studied in the form of convolution-type integrals, and more details on how to solve general gravity forward problems of sources with versatile geometries and density distributions using the Conv-Gauss-FFT algorithm are provided. In Sect. 6, many synthetic and real models are implemented to test the accuracy and efficiency of the presented algorithm. Section 7 provides a brief discussion on combining the Conv-Gauss-FFT algorithm with analytical solutions or some properly designed quadrature methods to evaluate the longrange and close-range contributions separately. Finally in Sect. 8, we draw some conclusions and suggest some possible future works.

2 A Hybrid Rectangle-Gaussian Grid for Continuous Convolutions

2.1 1D Continuous Convolutions

The 1D continuous convolution of two functions t and f, denoted as h = t * f, is written as:

$$h(x) = \int_{-\infty}^{\infty} f(\tilde{x})t(x - \tilde{x})d\tilde{x}.$$
 (1)

In the following, we call *t* the kernel function, *f* the source function and *h* the field function, based on their meanings in a forward modeling problem.

In a practical evaluation, the infinite integral usually reduces to a finite one due to the finite extent of the function *f*, indicating a finite length of the source body in a modeling problem:

$$h(x) = \int_{\tilde{x}_0}^{\tilde{x}_L} f(\tilde{x}) t(x - \tilde{x}) \mathrm{d}\tilde{x}.$$
 (2)

Now we use a hybrid rectangle-Gaussian grid to evaluate Eq. 2, that is, we divide the integration region $[\tilde{x}_0, \tilde{x}_{\tilde{L}}]$ into \tilde{L} equal subintervals of length $\Delta \tilde{x} = \frac{\tilde{x}_{\tilde{L}} - \tilde{x}_0}{L}$, after which we apply the Gaussian quadrature rule to each subinterval $[\tilde{x}_{\tilde{l}}, \tilde{x}_{\tilde{l}+1}]$, $\tilde{l} = 0, 1, 2, ..., \tilde{L} - 1$; thus, we have:

$$h(x) = \int_{\tilde{x}_{0}}^{\tilde{x}_{L}} f(\tilde{x})t(x-\tilde{x})d\tilde{x} = \sum_{\tilde{l}=0}^{\tilde{L}-1} \int_{\tilde{x}_{\tilde{l}}}^{\tilde{x}_{\tilde{l}+1}} f(\tilde{x})t(x-\tilde{x})d\tilde{x}$$

$$= \Delta \tilde{x} \sum_{\tilde{l}=0}^{\tilde{L}-1} \sum_{i=1}^{I} \mu_{i}^{(l)} f(\tilde{x}_{\tilde{l}+\tilde{\alpha}_{i}})t\left(x-\tilde{x}_{\tilde{l}+\tilde{\alpha}_{i}}\right),$$
(3)

where $\mu_i^{(l)}$ and $\tilde{\alpha}_i$ are the weights and nodes of a *I*-nodes Gauss–Legendre quadrature rule over the integration area [0, 1], and $\tilde{x}_{\tilde{l}+\tilde{\alpha}_i} = \tilde{x}_0 + (\tilde{l}+\tilde{\alpha}_i)\Delta \tilde{x}$.

Suppose that h(x) is needed on a size L regular grid: $x_l = x_0 + l\Delta x$, l = 0, 1, 2, ..., L - 1, by changing the order of the two summations in Eq. 3, we have:

$$h(x_l) = \Delta \tilde{x} \sum_{i=1}^{l} \mu_i^{(l)} \sum_{\tilde{l}=0}^{\tilde{L}-1} f(\tilde{x}_{\tilde{l}+\tilde{\alpha}_l}) t(x_l - \tilde{x}_{\tilde{l}+\tilde{\alpha}_l}).$$
(4)

If we choose identical grid interval for the source function f and the field function h, that is, $\Delta x = \Delta \tilde{x}$, then the value of the expression $x_l - \tilde{x}_{\tilde{l}+\tilde{\alpha}_i} = x_0 - \tilde{x}_0 + (l - \tilde{l} - \tilde{\alpha}_i)\Delta x$ depends solely on the value of $l - \tilde{l}$. Therefore, the kernel $t(x_l - \tilde{x}_{\tilde{l}+\tilde{\alpha}_i})$ becomes a function of $l - \tilde{l}$, and the second summation on the RHS of Eq. 4 becomes a discrete convolution.

Let

$$\mathbf{h} = [h(x_l)], \quad l = 0, 1, 2, \dots, L - 1$$
(5)

be the size L field vector containing values of the field function,

$$\mathbf{f}_{\tilde{\alpha}_i} = [f(\tilde{x}_{\tilde{l}+\tilde{\alpha}_i})], \quad \tilde{l} = 0, 1, 2, \dots, \tilde{L} - 1$$
(6)

be the size \tilde{L} source vector containing values of the source function on a $\tilde{\alpha}_i$ -shifted grid, and

$$\mathbf{t}_{\tilde{\alpha}_{i}} = [t(x_{l} - \tilde{x}_{\tilde{l} + \tilde{\alpha}_{i}})], \quad l - \tilde{l} = 1 - \tilde{L}, \dots, -1, 0, 1, \dots, L - 1$$
(7)

be the size $L + \tilde{L} - 1$ kernel vector containing values of the kernel function on a $\tilde{\alpha}_i$ -shifted grid, then Eq. 4 can be written in vector form as:

$$\mathbf{h} = \Delta x \sum_{i=1}^{I} \mu_i^{(I)} \mathbf{t}_{\tilde{\alpha}_i} \star \mathbf{f}_{\tilde{\alpha}_i}, \qquad (8)$$

where $\mathbf{t}_{\tilde{\alpha}_i} \star \mathbf{f}_{\tilde{\alpha}_i}$ denotes the discrete convolution:

$$[\mathbf{t}_{\tilde{\alpha}_i} \star \mathbf{f}_{\tilde{\alpha}_i}]_l = \sum_{\tilde{l}=0}^{\tilde{L}-1} [\mathbf{t}_{\tilde{\alpha}_i}]_{l-\tilde{l}} [\mathbf{f}_{\tilde{\alpha}_i}]_{\tilde{l}}, \quad l = 0, 1, \dots, L-1,$$
(9)

and it can be evaluated accurately and efficiently through the FFT algorithm. This will be discussed in detail in Sect. 3.

2.2 2D and 3D Continuous Convolutions

Applying a 2D or 3D hybrid rectangle-Gaussian grid, the above derivation can be easily extended to 2D and 3D continuous convolutions.

For a 2D finite continuous convolution:

$$h(x,z) = \int_{\tilde{x}_0}^{\tilde{x}_L} \int_{\tilde{z}_0}^{\tilde{z}_N} f(\tilde{x},\tilde{z}) t(x-\tilde{x},z-\tilde{z}) d\tilde{x} d\tilde{z},$$
(10)

we have:

$$\mathbf{h}_{2D} = \Delta x \Delta z \sum_{i=1}^{I} \sum_{k=1}^{K} \mu_i^{(I)} \omega_k^{(K)} \mathbf{t}_{\tilde{\alpha}_i \tilde{\gamma}_k} \star \mathbf{f}_{\tilde{\alpha}_i \tilde{\gamma}_k}, \qquad (11)$$

where $(\mu_i^{(I)}, \omega_k^{(K)})$ and $(\tilde{\alpha}_i, \tilde{\gamma}_k)$ are the weights and nodes of a $I \times K$ nodes 2D Gauss– Legendre quadrature rule over the integration area $[0, 1] \times [0, 1]$, and

$$\mathbf{h}_{2D} = [h(x_l, z_n)], l = 0, 1, 2, \dots, L-1, \quad n = 0, 1, 2, \dots, N-1$$
(12)

is the size $L \times N$ matrix containing values of the field function,

$$\mathbf{f}_{\tilde{\alpha}_{i}\tilde{\gamma}_{k}} = [f(\tilde{x}_{\tilde{l}+\tilde{\alpha}_{i}}, \tilde{z}_{\tilde{n}+\tilde{\gamma}_{k}})], \\ \tilde{l} = 0, 1, 2, \dots, \tilde{L} - 1, \quad \tilde{n} = 0, 1, 2, \dots, \tilde{N} - 1$$

$$(13)$$

is the size $\tilde{L} \times \tilde{N}$ matrix containing values of the source function on a $(\tilde{\alpha}_i, \tilde{\gamma}_k)$ -shifted grid, and

$$\mathbf{t}_{\tilde{\alpha}_{l}\tilde{\gamma}_{k}} = [t(x_{l} - \tilde{x}_{\tilde{l}+\tilde{\alpha}_{i}}, z_{n} - \tilde{z}_{\tilde{n}+\tilde{\gamma}_{k}})], l - \tilde{l} = 1 - \tilde{L}, \dots, -1, 0, 1, \dots, L - 1, n - \tilde{n} = 1 - \tilde{N}, \dots, -1, 0, 1, \dots, N - 1$$
(14)

is the size $(L + \tilde{L} - 1) \times (N + \tilde{N} - 1)$ matrix containing values of the kernel function on a $(\tilde{\alpha}_i, \tilde{\gamma}_k)$ -shifted grid.

Analogously, for a 3D finite continuous convolution:

$$h(x,y,z) = \int_{\tilde{x}_0}^{\tilde{x}_L} \int_{\tilde{y}_0}^{\tilde{y}_M} \int_{\tilde{z}_0}^{\tilde{z}_N} f(\tilde{x},\tilde{y},\tilde{z})t(x-\tilde{x},y-\tilde{y},z-\tilde{z})\mathrm{d}\tilde{x}\mathrm{d}\tilde{y}\mathrm{d}\tilde{z},\tag{15}$$

we have

🖄 Springer

$$\mathbf{h}_{3D} = \Delta_x \Delta_y \Delta_z \ \sum_{i=1}^{I} \sum_{j=1}^{J} \sum_{k=1}^{K} \mu_i^{(I)} v_j^{(J)} \omega_k^{(K)} \mathbf{t}_{\tilde{\alpha}_i \tilde{\beta}_j \tilde{\gamma}_k} \bigstar \mathbf{f}_{\tilde{\alpha}_i \tilde{\beta}_j \tilde{\gamma}_k}.$$
 (16)

where $(\mu_i^{(I)}, v_j^{(J)}, \omega_k^{(K)})$ and $(\tilde{\alpha}_i, \tilde{\beta}_j, \tilde{\gamma}_k)$ are the weights and nodes of a $I \times J \times K$ nodes 3D Gauss–Legendre quadrature rule over the integration volume $[0, 1]^3$, and

$$\mathbf{h}_{3D} = [h(x_l, y_m, z_n)], l = 0, 1, 2, \dots, L-1, \quad m = 0, 1, 2, \dots, M-1, \quad n = 0, 1, 2, \dots, N-1$$
(17)

is the size $L \times M \times N$ matrix containing values of the field function,

$$\mathbf{f}_{\tilde{\alpha}_{\tilde{i}}\tilde{\beta}_{\tilde{j}}\tilde{\gamma}_{\tilde{k}}} = [f(\tilde{x}_{\tilde{l}+\tilde{\alpha}_{\tilde{i}}}, \tilde{y}_{\tilde{m}+\tilde{\beta}_{\tilde{j}}}, \tilde{z}_{\tilde{n}+\tilde{\gamma}_{\tilde{k}}})], \\ \tilde{l} = 0, 1, 2, \dots, \tilde{L} - 1, \quad \tilde{m} = 0, 1, 2, \dots, \tilde{M} - 1, \quad \tilde{n} = 0, 1, 2, \dots, \tilde{N} - 1$$
(18)

is the size $\tilde{L} \times \tilde{M} \times \tilde{N}$ matrix containing values of the source function on a $(\tilde{\alpha}_i, \tilde{\beta}_j, \tilde{\gamma}_k)$ -shifted grid, and

$$\mathbf{t}_{\tilde{\alpha}_{i}\tilde{\beta}_{j}\tilde{\gamma}_{k}} = [t(x_{l} - \tilde{x}_{\tilde{l}+\tilde{\alpha}_{i}}, y_{m} - \tilde{y}_{\tilde{m}+\tilde{\beta}_{j}}, z_{n} - \tilde{z}_{\tilde{n}+\tilde{\gamma}_{k}})],$$

$$l - \tilde{l} = 1 - \tilde{L}, \dots, -1, 0, 1, \dots, L - 1,$$

$$m - \tilde{m} = 1 - \tilde{M}, \dots, -1, 0, 1, \dots, M - 1,$$

$$n - \tilde{n} = 1 - \tilde{N}, \dots, -1, 0, 1, \dots, N - 1$$
(19)

is the size $(L + \tilde{L} - 1) \times (M + \tilde{M} - 1) \times (N + \tilde{N} - 1)$ matrix containing values of the kernel function on a $(\tilde{\alpha}_i, \tilde{\beta}_j, \tilde{\gamma}_k)$ -shifted grid.

3 Discrete Convolutions Via the FFT Algorithm

We have reduced the 1D, 2D and 3D continuous convolutions to the weighted sum of a series of discrete convolutions using 1D, 2D or 3D hybrid rectangle-Gaussian grids. It can be observed from Eqs. 8, 11 and 16 that the efficiency of the algorithm relies on the fast calculation of discrete convolutions. It is well known that discrete convolutions can be evaluated efficiently and accurately, to almost machine precision, through the FFT algorithm (Vogel 2002). In the following, we provide some basic concepts of the relationships between discrete convolutions, Toeplitz systems and the FFT algorithm.

The derivation here is a bit different from those provided in Vogel (2002) in several aspects: (1) we deal with the more general case when the vectors **h** and **f** are of the same or different sizes L and \tilde{L} , respectively. The size of the kernel vector **t** is then $L + \tilde{L} - 1$, as has been discussed above; (2) we add no extra zeros when embedding the Toeplitz matrix T into its circulant extension; and (3) we use slightly different symbols to simplify expressions.

The 1D discrete convolution of a source vector $\mathbf{f} = (f_0, f_1, \dots, f_{\tilde{L}-1})$ and a kernel vector $\mathbf{t} = (t_{1-\tilde{L}}, t_{2-\tilde{L}}, \dots, t_{-1}, t_0, t_1, \dots, t_{L-2}, t_{L-1})$ is written as:

$$\mathbf{h}]_{l} = [\mathbf{t} \star \mathbf{f}]_{l} = \sum_{\bar{l}=0}^{\tilde{L}-1} t_{l-\bar{l}} f_{\bar{l}}, \quad l = 0, 1, \dots, L-1.$$
(20)

Let

ſ

$$T = \begin{bmatrix} t_0 & t_{-1} & \dots & t_{2-\bar{L}} & t_{1-\bar{L}} \\ t_1 & t_0 & t_{-1} & \ddots & t_{2-\bar{L}} \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ t_{L-2} & \ddots & t_{L-\bar{L}+1} & t_{L-\bar{L}} & t_{L-\bar{L}-1} \\ t_{L-1} & t_{L-2} & \dots & t_{L-\bar{L}+1} & t_{L-\bar{L}} \end{bmatrix},$$
(21)
$$\mathbf{f} = \begin{bmatrix} f_0 \\ f_1 \\ \vdots \\ f_{\bar{L}-2} \\ f_{\bar{L}-1} \end{bmatrix}, \quad \mathbf{h} = \begin{bmatrix} h_0 \\ h_1 \\ \vdots \\ h_{L-2} \\ h_{L-1} \end{bmatrix},$$
(22)

then the discrete convolution can be written in matrix-vector product form as:

$$\mathbf{h} = T\mathbf{f},\tag{23}$$

where $T = \text{toeplitz}(\mathbf{t})$.

Toeplitz matrix–vector products can be efficiently computed by combining circulant embedding with FFTs (Vogel 2002). Denote $C = \text{circulant}(\mathbf{c})$ as the $n \times n$ circulant matrix

	c_0	c_{n-1}		c_2	c_1		
	c_1	c_0	c_{n-1}	·.	<i>c</i> ₂		
C =	:	c_1	c_0	·.	÷	,	(24)
	c_{n-2}	·	·	·	c_{n-1}		
	c_{n-1}	c_{n-2}		c_1	c_0		

with $\mathbf{c} = (c_0, c_1, \dots, c_{n-1})$. The Toeplitz matrix T can be embedded in a size $(L + \tilde{L} - 1) \times (L + \tilde{L} - 1)$ circulant matrix $\hat{T} = \text{circulant}(\hat{\mathbf{t}})$, with $\hat{\mathbf{t}} = (t_0, t_1, \dots, t_{L-1}, t_{1-\tilde{L}}, \dots, t_{L-\tilde{L}-2}, t_{L-\tilde{L}-1})$. Therefore, the Toeplitz matrix–vector product $T\mathbf{f}$ can be embedded in a circulant matrix–vector product $\hat{T}\mathbf{f}$ and then evaluated through the FFT algorithm:

$$\hat{\mathbf{h}} = \hat{T}\hat{\mathbf{f}} = \hat{T}\begin{bmatrix}\mathbf{f}\\\mathbf{0}_{(L-1)\times 1}\end{bmatrix} = \begin{bmatrix}T\mathbf{f}\\A\mathbf{f}\end{bmatrix} = \mathbf{F}^{-1}\Big[\mathbf{F}(\hat{\mathbf{f}})\mathbf{F}(\hat{\mathbf{t}})\Big],$$

$$\mathbf{h} = [\hat{\mathbf{h}}]_{L\times 1},$$
(25)

where **F** and **F**⁻¹ represent the 1D forward and inverse discrete Fourier transform operators, respectively. The vector $\hat{\mathbf{f}} = (f_0, f_1, \dots, f_{\tilde{L}-1}, \underbrace{0, \dots, 0}_{L-1})$ is obtained by a zero-padding

🖄 Springer

extension of the original source vector \mathbf{f} , and $\hat{\mathbf{t}} = \operatorname{circshift}(\mathbf{t}, L)$ is obtained by circularly shifting the elements in the original kernel vector \mathbf{t} by *L* positions. Finally, the field vector \mathbf{h} can be obtained by extracting the first *L* elements in the extended field vector $\hat{\mathbf{h}}$.

Analogously, the derivation above can be extended to the evaluation of 2D and 3D discrete convolutions using the 2D and 3D FFT algorithms.

For the 2D discrete convolution:

$$[\mathbf{h}]_{l,n} = [\mathbf{t} \star \mathbf{f}]_{l,n} = \sum_{\tilde{l}=0}^{\tilde{L}-1} \sum_{\tilde{n}=0}^{\tilde{N}-1} t_{l-\tilde{l},n-\tilde{n}} f_{\tilde{l},\tilde{n}},$$

$$l = 0, 1, \dots, L-1, \quad n = 0, 1, \dots, N-1,$$
(26)

the algorithm can be written as:

$$\hat{\mathbf{h}}_{2D} = \mathbf{F}^{-1} \Big[\mathbf{F}(\hat{\mathbf{f}}_{2D}) \mathbf{F}(\hat{\mathbf{t}}_{2D}) \Big],$$

$$\mathbf{h}_{2D} = [\hat{\mathbf{h}}_{2D}]_{L \times N}.$$
(27)

Here **F** and **F**⁻¹ represent the 2D forward and inverse discrete Fourier transform operators, respectively. The extended source matrix $\hat{\mathbf{f}}_{2D}$ is obtained by a zero-padding extension of the original source matrix $[f_{\bar{l},\bar{n}}]$ of size $\tilde{L} \times \tilde{N}$ to the size $(L + \tilde{L} - 1) \times (N + \tilde{N} - 1)$, and $\hat{\mathbf{t}}_{2D}$ = circshift(\mathbf{t}_{2D}, L, N) is obtained by circularly shifting the elements in the original kernel matrix $[t_{l-\tilde{l},n-\tilde{n}}]$ by *L* and *N* positions along two dimensions, respectively. Finally, the field matrix \mathbf{h}_{2D} can be obtained by extracting the first $L \times N$ elements in the extended field matrix $\hat{\mathbf{h}}_{2D}$.

Similarly, the algorithm for the 3D discrete convolution:

$$[\mathbf{h}]_{l,m,n} = [\mathbf{t} \star \mathbf{f}]_{l,m,n} = \sum_{\tilde{l}=0}^{\tilde{L}-1} \sum_{\tilde{m}=0}^{\tilde{M}-1} \sum_{\tilde{n}=0}^{\tilde{N}-1} t_{l-\tilde{l},m-\tilde{m},n-\tilde{m}} f_{\tilde{l},\tilde{m},\tilde{n}},$$

$$l = 0, 1, \dots, L-1, \quad m = 0, 1, \dots, M-1, \quad n = 0, 1, \dots, N-1,$$
(28)

is written as:

$$\hat{\mathbf{h}}_{3D} = \mathbf{F}^{-1} \Big[\mathbf{F}(\hat{\mathbf{f}}_{3D}) \mathbf{F}(\hat{\mathbf{t}}_{3D}) \Big],$$

$$\mathbf{h}_{3D} = [\hat{\mathbf{h}}_{3D}]_{L \times M \times N},$$
(29)

where **F** and **F**⁻¹ represent the 3D forward and inverse discrete Fourier transform operators, respectively. The extended 3D source matrix $\hat{\mathbf{f}}_{3D}$ is obtained by a zero-padding extension of the original 3D source matrix $[f_{\tilde{l},\tilde{m},\tilde{n}}]$ of size $\tilde{L} \times \tilde{M} \times \tilde{N}$ to the size $(L + \tilde{L} - 1) \times (M + \tilde{M} - 1) \times (N + \tilde{N} - 1)$, and $\hat{\mathbf{t}}_{3D}$ = circshift(\mathbf{t}_{3D}, L, M, N) is obtained by circularly shifting the elements in the original kernel matrix $[t_{l-\tilde{l},m-\tilde{m},n-\tilde{n}}]$ by L, M and Npositions along three dimensions, respectively. Finally, the 3D field matrix \mathbf{h}_{3D} can be obtained by extracting the first $L \times M \times N$ elements in the extended 3D field matrix $\hat{\mathbf{h}}_{3D}$.

4 The Conv-Gauss-FFT Algorithm

Based on the derivations in the previous two sections, we introduce here an efficient algorithm, namely the Conv-Gauss-FFT algorithm, for the calculation of continuous convolution-type integrals by a combined use of hybrid rectangle-Gaussian grids and the FFT algorithm.

For 1D continuous convolutions, by combining Eqs. 8 and 25, we have:

$$\hat{\mathbf{h}} = \Delta x \sum_{i=1}^{I} \mu_i^{(I)} \mathbf{F}^{-1} \Big[\mathbf{F}(\hat{\mathbf{f}}_{\tilde{\alpha}_i}) \mathbf{F}(\hat{\mathbf{t}}_{\tilde{\alpha}_i}) \Big]$$

= $\Delta x \mathbf{F}^{-1} \left[\sum_{i=1}^{I} \mu_i^{(I)} \mathbf{F}(\hat{\mathbf{f}}_{\tilde{\alpha}_i}) \mathbf{F}(\hat{\mathbf{t}}_{\tilde{\alpha}_i}) \Big],$
$$\mathbf{h} = [\hat{\mathbf{h}}]_{I \times 1}.$$
 (30)

We call this approach the 1D Conv-Gauss-FFT algorithm.

The 2D and 3D Conv-Gauss-FFT algorithms for the evaluations of the 2D and the 3D continuous convolutions can be obtained analogously. Combining Eqs. 11 and 27, we have the 2D Conv-Gauss-FFT algorithm:

$$\hat{\mathbf{h}}_{2D} = \Delta_{x} \Delta_{z} \mathbf{F}^{-1} \left[\sum_{i=1}^{I} \sum_{k=1}^{K} \mu_{i}^{(I)} \omega_{k}^{(K)} \mathbf{F}(\hat{\mathbf{f}}_{\tilde{a}_{i}\tilde{\gamma}_{k}}) \mathbf{F}(\hat{\mathbf{t}}_{\tilde{a}_{i}\tilde{\gamma}_{k}}) \right],$$

$$\mathbf{h}_{2D} = [\hat{\mathbf{h}}_{2D}]_{I \times N}.$$
(31)

Similarly, by combining Eqs. 16 and 29, we have the 3D Conv-Gauss-FFT algorithm:

$$\hat{\mathbf{h}}_{3D} = \Delta_x \Delta_y \Delta_z \, \mathbf{F}^{-1} \left[\sum_{i=1}^{I} \sum_{j=1}^{J} \sum_{k=1}^{K} \mu_i^{(I)} v_j^{(J)} \omega_k^{(K)} \mathbf{F}(\hat{\mathbf{f}}_{\tilde{\alpha}_i \tilde{\beta}_j \tilde{\gamma}_k}) \mathbf{F}(\hat{\mathbf{t}}_{\tilde{\alpha}_i \tilde{\beta}_j \tilde{\gamma}_k}) \right],$$

$$\mathbf{h}_{3D} = [\hat{\mathbf{h}}_{3D}]_{L \times M \times N}.$$
(32)

The complexity of the algorithm depends on several parameters, including the source grid size, the field grid size, and the number of Gaussian nodes used in each direction. Take the 3D Conv-Gauss-FFT algorithm for instance, the numerical complexity of the algorithm is $O(IJKS \log S)$, with I, J, K the number of Gaussian nodes along three directions, and $S = (L + \tilde{L} - 1) \times (M + \tilde{M} - 1) \times (N + \tilde{N} - 1)$ is determined by the sizes of both the source grid $(\tilde{L} \times \tilde{M} \times \tilde{N})$ and the field grid $(L \times M \times N)$. The total number of discrete Fourier transforms needed is 2IJK + 1.

The accuracy of the algorithm also depends on several aspects, among which the wellbehavedness of the source and the kernel functions, i.e., how well can continuous polynomials approximate these functions, plays the central role. When the source and the kernel functions are well-behaved, that is, smooth enough and with no singularity, the algorithm can achieve high accuracy as long as a sufficiently large number of Gaussian nodes are used. However, when the source function is only piecewise continuous, as is usually the case in a gravity forward problem, the density function changes discontinuously across the boundary of the source body, the algorithm may introduce some observable errors for field points close to the source's boundary. Furthermore, when the kernel function has a strong singularity, as is the case for high-order derivatives of Newton's integral kernel, the algorithm fails if the convolution integral becomes singular and cannot be properly solved using a regular quadrature method, including the hybrid rectangle-Gaussian quadrature method used here.

The Conv-Gauss-FFT algorithm, including the 2D and 3D cases, has been coded in MATLAB language. Together with a point-in-polygon algorithm and a point-in-polyhedron algorithm, the method can be used to solve general 2D and 3D gravity forward problems including sources with arbitrary geometry and arbitrary density distribution. Moreover, the algorithm may also be useful in solving other convolution-type integrals in geophysics and geodesy (Hirt et al. 2011). The efficiency, accuracy and robustness of the algorithm will be assessed by numerical comparisons with examples derived from the literature.

5 General Gravity Forward Modeling Using the Conv-Gauss-FFT Algorithm

5.1 The 2D Forward Problem

The general 2D gravity forward problem can be written as a continuous convolution-type integral:

$$\frac{\partial^{i+k}}{\partial x^i \partial z^k} \phi(x, z) = 2G \int_{\tilde{x}_0}^{\tilde{x}_{\tilde{L}}} \int_{\tilde{z}_0}^{\tilde{z}_{\tilde{N}}} \rho(\tilde{x}, \tilde{z}) \frac{\partial^{i+k}}{\partial x^i \partial z^k} \left[\log \frac{1}{\mathbf{d}(x - \tilde{x}, z - \tilde{z})} \right] \mathrm{d}\tilde{x} \mathrm{d}\tilde{z}, \tag{33}$$

where G is the gravitational constant, $\rho(\tilde{x}, \tilde{z})$ is the 2D density distribution within the integration region, ϕ is the gravity potential and $\frac{\partial^{i+k}}{\partial x^i \partial z^k}$ represent its derivative components, and

$$d(x - \tilde{x}, z - \tilde{z}) = \sqrt{(x - \tilde{x})^2 + (z - \tilde{z})^2}$$
(34)

is the 2D Euclidean distance between a field point (x, z) and a source point (\tilde{x}, \tilde{z}) .

Figure 1 shows some basic concepts of the 2D Conv-Gauss-FFT algorithm. The region covered by filed points (the red grid) is denoted as:

$$\Omega_{\text{field}} = \{ (x, z) | x_0 \le x \le x_{L-1}, z_0 \le z \le z_{N-1} \}.$$
(35)

The source region (the green grid) is defined as:

$$\Omega_{\text{source}} = \left\{ (\tilde{x}, \tilde{z}) | \tilde{x}_{0} \leq \tilde{x} \leq \tilde{x}_{\tilde{L}}, \tilde{z}_{0} \leq \tilde{z} \leq \tilde{z}_{\tilde{N}} \right\},
\tilde{x}_{0} = x_{0} + \left\lfloor \frac{\tilde{x}_{\min} - x_{0}}{\Delta x} \right\rfloor \Delta x, \quad d\tilde{x}_{\tilde{L}} = x_{0} + \left\lceil \frac{\tilde{x}_{\max} - x_{0}}{\Delta x} \right\rceil \Delta x,
\tilde{z}_{0} = z_{0} + \left\lfloor \frac{\tilde{z}_{\min} - z_{0}}{\Delta z} \right\rfloor \Delta z, \quad d\tilde{z}_{\tilde{N}} = z_{0} + \left\lceil \frac{\tilde{z}_{\max} - z_{0}}{\Delta z} \right\rceil \Delta z,$$
(36)

where $\tilde{x}_{\min}, \tilde{x}_{\max}, \tilde{z}_{\min}, \tilde{z}_{\max}$ are the upper and lower limits of the coordinate values of the source polygon's (the blue polygon's) vertices, and the symbols $\lfloor \rfloor$ and $\lceil \rceil$ represent the floor and ceiling functions, respectively.

We mention here that although we have drawn the source grid contained within the field grid in Fig. 1, there is no limitation on the geometrical relationship between these two regions. They can either be separated $(\Omega_{\text{field}} \cap \Omega_{\text{source}} = \emptyset)$, overlapped $(\Omega_{\text{field}} \cap \Omega_{\text{source}} \neq \emptyset)$, or with one contained in another $(\Omega_{\text{field}} \cap \Omega_{\text{source}} = \Omega_{\text{field}} \text{ or } \Omega_{\text{source}})$.

Springer



Fig. 1 Illustration of the 2D gravity forward model based on the 2D Conv-Gauss-FFT algorithm

However, to make the Conv-Gauss-FFT algorithm work, the grid intervals of these two regions along each direction must be the same, that is, the condition $\Delta \tilde{x} = \Delta x, \Delta \tilde{y} = \Delta y, \Delta \tilde{z} = \Delta z$ is met.

Now we use the 2D Conv-Gauss-FFT algorithm (Eq. 31) to evaluate Eq. 33. Obviously, the kernel function t can be evaluated analytically for the gravity potential and its derivatives of arbitrary order. Taking the vertical component of the gravity vector for instance, we have:

$$t(x-\tilde{x},z-\tilde{z}) = \frac{\partial}{\partial z} \left[\log \frac{1}{\mathbf{d}(x-\tilde{x},z-\tilde{z})} \right] = \frac{-(z-\tilde{z})}{(x-\tilde{x})^2 + (z-\tilde{z})^2}.$$
 (37)

Then for each pair of shift parameters $(\tilde{\alpha}_i, \tilde{\gamma}_k)$, the kernel matrix $\hat{\mathbf{t}}_{\tilde{\alpha}_i \tilde{\gamma}_k}$ can be obtained by circularly shifting the elements in the original kernel matrix $\mathbf{t}_{\tilde{\alpha}_i \tilde{\gamma}_k}$ by *L* and *N* positions along two dimensions, respectively, with $\mathbf{t}_{\tilde{\alpha}_i \tilde{\gamma}_k}$ defined in Eq. 14.

The kernel functions correspond to the gravity vector, and the gravity gradient tensor components caused by a 2D source body are written as:

$$[\phi_x, \phi_z] \longleftrightarrow \left[\frac{-(x - \tilde{x})}{d^2}, \frac{-(z - \tilde{z})}{d^2} \right],$$
(38)

$$[\phi_{xx}, \phi_{xz}, \phi_{zz}] \longleftrightarrow \left[\frac{2(x-\tilde{x})^2 - d^2}{d^4}, \frac{2(x-\tilde{x})(z-\tilde{z})}{d^4}, \frac{2(z-\tilde{z})^2 - d^2}{d^4} \right].$$
(39)

Similarly, for each pair of shift parameters $(\tilde{\alpha}_i, \tilde{\gamma}_k)$, the source matrix $\mathbf{f}_{\tilde{\alpha}_i \tilde{\gamma}_k}$ is obtained by the zero-padding extension of the original source matrix $\mathbf{f}_{\tilde{\alpha}_i \tilde{\gamma}_k}$, which is defined in Eq. 13 by the source function $f(\tilde{x}_{\tilde{l}+\tilde{\alpha}_i}, \tilde{z}_{\tilde{n}+\tilde{\gamma}_k})$. For the polygonal model in Fig. 1, values of the source function can be determined by both the density distribution $\rho(\tilde{x}, \tilde{z})$ within the polygon and a point-in-polygon algorithm. Denote the region covered by the source polygon as Ω_{polygon} , we then have:

$$f(\tilde{x}, \tilde{z}) = \begin{cases} \rho(\tilde{x}, \tilde{z}), & \text{if } (\tilde{x}, \tilde{z}) \in \Omega_{\text{polygon}}, \\ 0, & \text{if } (\tilde{x}, \tilde{z}) \notin \Omega_{\text{polygon}}. \end{cases}$$
(40)

As shown in Fig. 1, the source function is sampled on a hybrid rectangle-Gaussian grid (cross symbols) with the number of Gaussian nodes chosen as I = K = 2. For each pair of shift parameters $(\tilde{\alpha}_i, \tilde{\gamma}_k)$, the point-in-polygon algorithm is used to determine whether a grid point lies inside (blue cross symbols) or outside of (green cross symbols) the source polygon. For the point-in-polygon test, we use the MATLAB function inpoly.m downloaded from the MATLAB central. The function is provided by Darren Engwirda at http://cn.mathworks.com/matlabcentral/fileexchange/10391-fast-points-in-polygon-test.

In addition to polygonal models, the source region can also be defined by analytical functions. Martin-Atienza and Garcia-Abdeslem (1999) studied 2D gravity modeling with source body bounded either laterally by functions of depth:

$$\Omega_{2\mathrm{D}} = \{ (\tilde{x}, \tilde{z}) \colon h_1(\tilde{z}) \le \tilde{x} \le h_2(\tilde{z}), \tilde{z}_1 \le \tilde{z} \le \tilde{z}_2 \},\tag{41}$$

or vertically by functions of the horizontal position:

$$\Omega_{2\mathrm{D}} = \{ (\tilde{x}, \tilde{z}) \colon \tilde{x}_1 \le \tilde{x} \le \tilde{x}_2, g_1(\tilde{x}) \le \tilde{z} \le g_2(\tilde{x}) \}.$$

$$(42)$$

For these analytically defined geometries, it is simpler to determine whether a source point lies within the source region or not. The point-in-polygon algorithm is no longer necessary; a simple comparison to the two analytical boundary functions would immediately tell.

5.2 The 3D Forward Problem

We have discussed in detail the 2D Conv-Gauss-FFT solution for a general 2D gravity forward problem. The 3D forward problem can be solved in a very similar way based on the 3D Conv-Gauss-FFT algorithm (Eq. 32) and a point-in-polyhedron algorithm.

The general 3D gravity forward problem can be written as:

$$\frac{\partial^{i+j+k}}{\partial x^{i}\partial y^{j}\partial z^{k}}\phi(x,y,z) = G \int_{\tilde{x}_{0}}^{\tilde{x}_{L}} \int_{\tilde{y}_{0}}^{\tilde{y}_{M}} \int_{\tilde{z}_{0}}^{\tilde{z}_{N}} \rho(\tilde{x},\tilde{y},\tilde{z}) \frac{\partial^{i+j+k}}{\partial x^{i}\partial y^{j}\partial z^{k}} \left[\frac{1}{r(x-\tilde{x},y-\tilde{y},z-\tilde{z})}\right] d\tilde{x}d\tilde{y}d\tilde{z},$$
(43)

where

$$r(x - \tilde{x}, y - \tilde{y}, z - \tilde{z}) = \sqrt{(x - \tilde{x})^2 + (y - \tilde{y})^2 + (z - \tilde{z})^2}$$
(44)

is the 3D Euclidean distance between a field point (x, y, z) and a source point $(\tilde{x}, \tilde{y}, \tilde{z})$, and $\rho(\tilde{x}, \tilde{y}, \tilde{z})$ is the 3D density distribution within the integration volume.

Again, kernel functions of the gravity potential and its derivatives of arbitrary order can be obtained analytically. The kernels correspond to the most frequently used gravity vector, and gravity gradient tensor components are:

$$\left[\phi_x \phi_y \phi_z\right] \longleftrightarrow \left[\frac{-(x-\tilde{x})}{r^3} \frac{-(y-\tilde{y})}{r^3} \frac{-(z-\tilde{z})}{r^3}\right],\tag{45}$$

$$\begin{bmatrix} \phi_{xx} \phi_{yy} \phi_{zz} \\ \phi_{xy} \phi_{xz} \phi_{yz} \end{bmatrix} \longleftrightarrow \begin{bmatrix} \frac{3(x-\tilde{x})^2 - r^2}{r^5} \frac{3(y-\tilde{y})^2 - r^2}{r^5} \frac{3(z-\tilde{z})^2 - r^2}{r^5} \\ \frac{3(x-\tilde{x})(y-\tilde{y})}{r^5} \frac{3(x-\tilde{x})(z-\tilde{z})}{r^5} \frac{3(y-\tilde{y})(z-\tilde{z})}{r^5} \end{bmatrix}.$$
 (46)

Clearly all kernel functions listed above have very simple expressions and can be evaluated very quickly with almost negligible computational cost.

Analogous to the 2D case, for each group of shift parameters $(\tilde{\alpha}_i, \beta_j, \tilde{\gamma}_k)$, the point-inpolyhedron algorithm is used to determine whether a grid point lies within or out of the



Fig. 2 A point-in-polyhedron test of a regular grid of points using an Octahedron

source polyhedron. Figure 2 shows a point-in-polyhedron test of a regular grid of points using an Octahedron source body. The MATLAB function inpolyhedron.m downloaded from the MATLAB central provided by Sven Holcombe is used for the test. Denote the region occupied by the source polyhedron as $\Omega_{polyhedron}$, we then have:

$$f(\tilde{x}, \tilde{y}, \tilde{z}) = \begin{cases} \rho(\tilde{x}, \tilde{y}, \tilde{z}), & \text{if } (\tilde{x}, \tilde{y}, \tilde{z}) \in \Omega_{\text{polyhedron}}, \\ 0, & \text{if } (\tilde{x}, \tilde{y}, \tilde{z}) \notin \Omega_{\text{polyhedron}}. \end{cases}$$
(47)

In addition to polyhedral bodies, a 3D source region can also be described by a layer with uneven top and bottom surfaces, like those frequently used in a terrain correction problem (Parker 1973; Wu 2016):

$$\Omega_{3D} = \{ (\tilde{x}, \tilde{y}, \tilde{z}) \colon \tilde{x}_1 \le \tilde{x} \le \tilde{x}_2, \, \tilde{y}_1 \le \tilde{y} \le \tilde{y}_2, \, \tilde{z}_1(\tilde{x}, \tilde{y}) \le \tilde{z} \le \tilde{z}_2(\tilde{x}, \tilde{y}) \}.$$
(48)

The top and bottom surfaces can be step functions, piecewise triangular interpolated functions, or smooth functions as discussed in Wu (2016). When these topographic models are considered, the point-in-polyhedron algorithm is no longer necessary, and a simple comparison to the top and bottom surface functions would directly tell if a point lies within or outside the source region.

6 Numerical Tests

In this section, we use both 2D and 3D synthetic and real models, with versatile geometries, constant or variable density distributions, to test the Conv-Gauss-FFT algorithm. For each model, space-domain analytical solutions are used as precise references. Numerical accuracy and computational efficiency of the Conv-Gauss-FFT algorithm are then tested by comparing to space-domain solutions.

For comparison of numerical accuracy, we provide difference maps between the Conv-Gauss-FFT algorithm and space-domain analytical solutions. Alternatively, we also use several different criteria, including the absolute error, the relative error and the relative root-mean-square (rrms) error. The absolute error and the relative error are used when comparison is made at a single field point, e.g., at the origin of the coordinate system. The relative root-mean-square error is used for comparison of fields on profiles or on the whole 2D or 3D regular grid.

We also include the computing times of both algorithms for all synthetic and real models. We use T_{space} as the computing time for space-domain methods and T_{CGF} for the Conv-Gauss-FFT algorithm. It should be mentioned that in the numerical examples we provided, T_{space} usually represents the time cost of the space-domain solution calculating gravity anomalies at one single point or on one single profile, while T_{CGF} usually represents the time cost of the Conv-Gauss-FFT algorithm calculating gravity anomalies on the whole 2D or 3D grid. The algorithms were coded in MATLAB language and executed on a Intel core PC with 8 GB of RAM and a i5-3470 CPU having a clock speed of 3.20 GHz.

6.1 2D Examples

First we test our algorithm using a simple 2D model, an infinitely extended horizontal cylinder. The cylinder is of uniform density contrast $\rho = 1000$ kg m⁻³. It is centered at the origin of the coordinate system, and its radius is a = 10 km. Field points on a regular grid x = -20:1:20 km, z = -20:1:20 km are evaluated using both methods. Figure 3 compares



Fig. 3 a Space-domain forward gravity anomaly ϕ_z (mGal) due to an infinitely extended horizontal cylinder and the associated error maps of the 2D Conv-Gauss-FFT algorithm with different number of Gaussian nodes, **b** I = K = 10, **c** I = K = 20, **d** I = K = 30

forward results of the gravity component ϕ_z caused by the cylinder using both spacedomain analytical solution and the 2D Conv-Gauss-FFT algorithm with different numbers of Gaussian nodes.

Space-domain solution of the ϕ_z anomaly due to this model is easily obtained:

$$\phi_{z}(x,z) = \begin{cases} -2G\frac{\pi a^{2}\rho z}{x^{2}+z^{2}}, & \text{if } (x,z) \notin \Omega_{\text{cylinder}}, \\ -2G\pi\rho z, & \text{if } (x,z) \in \Omega_{\text{cylinder}}, \end{cases}$$
(49)

where Ω_{cylinder} denotes the area covered by the cylinder: $\Omega_{\text{cylinder}} = \{(x, z) | x^2 + z^2 \le a^2\}$. Figure 3a shows space-domain forward results of the ϕ_z anomaly. As indicated in Eq. 49, the ϕ_z anomaly is symmetric with respect to the *x*-axis and antisymmetric with respect to the *z* axis. Inside the cylinder, it is linearly dependent on the value of *z*.

Figure 3b–d shows the associated error maps of the 2D Conv-Gauss-FFT algorithm with numbers of Gaussian nodes I = K = 10, 20, 30. For most part of the field region, the Conv-Gauss-FFT algorithm with a small number of Gaussian nodes can already generate forward results almost identical to the space-domain solutions. However, the difference between these two algorithms becomes noticeable near the boundary of the cylinder, and it cannot be fully suppressed by increasing the number of Gaussian nodes. The reason for this, as has been discussed before, is that the density function changes discontinuously across the boundary of the source body, producing a discontinuous source function that cannot be well approximated by continuous polynomials. The space-domain solution costs only Figure 4 provides more details of the differences between these two algorithms. Figure 4a plots the absolute errors of the I = K = 30 Conv-Gauss-FFT algorithm against the point–cylinder distance, i.e., the minimum distance from each field point to the boundary of the cylinder: $\delta = |\sqrt{x^2 + z^2} - a|$. Apparently the point–cylinder distance in general controls the upper bound of the error. The larger the distance, the smaller the error's upper bound. Figure 4b shows histogram of the error. Most of the errors lie within the range [-0.01, 0.01] mGal. Figure 4c shows absolute error changes at five randomly picked field points: *A*, *B*, *C*, *D* and *E* with increasing number of Gaussian nodes. In general, the errors decrease with increasing number of Gaussian nodes. However, they do not decrease monotonically due to unknown reasons. Figure 4d shows the overall rrms errors of the



Fig. 4 Absolute errors of the I = K = 30 Conv-Gauss-FFT algorithm against the point–cylinder distance, **b** histogram of the error, **c** absolute error changes with increasing number of Gaussian nodes at five field points, **d** the overall rrms error of the Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes for the modeling of the ϕ_x and ϕ_z components

Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes for the modeling of the ϕ_x and ϕ_z components on the whole 2D grid. These two components have exactly the same rrms error because both their analytical expressions and kernel functions are mathematically equivalent. The changing of rrms errors has similar behaviors with those observed in Fig. 4c. In general, they also decrease, but not monotonically, with increasing number of Gaussian nodes.

Next we test our method using a variable density model, a 2D prism with linear depthdependent density function $\rho(\tilde{z}) = 100 + 60\tilde{z}$ kg m⁻³. As shown in Fig. 5, the prism extends from - 5 to 5 km horizontally and 0 to 15 km vertically. For the Conv-Gauss-FFT algorithm, a 2D grid of field points: x = -20:1:20 km, z = -20:1:20 km are calculated. For the space-domain analytical solution, only five profiles A, B, C, D and E at different heights z = -1 km (solid line), z = -2 km (dashed line), z = -4 km (dotted line), z = -8 km (cross), and z = -16 km (asterisk) in the upper half-space are evaluated.



Fig. 5 A 2D prism model with linear depth-dependent density function $\rho = 100 + 60\tilde{z}$ kg m⁻³. Five field profiles A–E at different heights z = -1, -2, -4, -8, -16 km in the upper half-space are calculated and compared using space-domain analytical solutions and the 2D Conv-Gauss-FFT algorithm

Therefore, the comparison between these two methods is carried out only on these five profiles.

Unlike the horizontal cylinder model we have studied above, the geometry of the 2D prism model chosen here fits perfectly with the rectangle-Gaussian grid used in the Conv-Gauss-FFT algorithm. As a result, the density function changes smoothly within each small rectangle area, leading to integral kernels that can be well approximated by continuous polynomials, which then ensures the high accuracy of the Gaussian quadrature rule.

Figure 6 compares space-domain and Conv-Gauss-FFT forward results of the gravity vector component ϕ_z and the gravity gradient tensor component ϕ_{zz} on the five profiles *A*, *B*, *C*, *D*, *E* listed above caused by the linear variable density model in Fig. 5. It can be observed that for both the ϕ_z and ϕ_{zz} components, the Conv-Gauss-FFT algorithm agrees very well, to almost machine precision, to the space-domain analytical solutions. The Conv-Gauss-FFT algorithm converges faster for profiles higher above the source because gravity anomalies generated farther from the source are smoother and can be better approximated by lower-order polynomials. The space-domain solution costs about 0.0015 s modeling these five profiles, while the Conv-Gauss-FFT algorithm with I = K = 10 costs about 0.09 s modeling the whole 2D grid.

Finally, we test our algorithm using four 2D models in the literature. They have been studied by several authors using versatile variable density distributions. Figure 7 compares space-domain analytical solutions and the Conv-Gauss-FFT forward results of the ϕ_z anomaly caused by these four models.

Figure 7a shows forward results of a 2D rectangular prism, which extends vertically from 1 to 2 km, and horizontally from 3 to 9 km; it has been studied by Rao (1986), Zhang et al. (2001) and D'Urso (2015), by assuming a depth-dependent density function

$$\rho(\tilde{z}) = 1.54 + 0.24\tilde{z} - 0.035\tilde{z}^2.$$
⁽⁵⁰⁾

Space-domain results are extracted from table 1 in D'Urso (2015). The Conv-Gauss-FFT algorithm with I = K = 2 is applied to calculate a 2D grid of field points: x = 0.0.01:12 km, z = 0.1:3 km, after which field values on the profile z = 0 km with horizontal positions coincide with those in D'Urso (2015) are extracted for comparison. The Conv-Gauss-FFT algorithm costs about 0.02 s evaluating the whole 2D grid.

Figure 7b shows forward results of a 2D polygon; it has been studied by Garcia-Abdeslem et al. (2005), Zhou (2008) and D'Urso (2015). It refers to the Sebastián Vizcaíno Basin in Mexico for which the density contrast has been assumed in the form

$$\rho(\tilde{z}) = -0.7 + 0.2548\tilde{z} - 0.0273\tilde{z}^2.$$
⁽⁵¹⁾

Space-domain results are extracted from table 2 in D'Urso (2015). The Conv-Gauss-FFT algorithm with I = 2, K = 4 is applied to calculate gravity anomalies on the 2D grid: x = -0.4:0.01:2.2 km, z = 0:0.05:3 km, after which field values on the profile z = 0 km with horizontal positions coincide with those in D'Urso (2015) are extracted for comparison. The Conv-Gauss-FFT algorithm costs about 0.05 s evaluating the whole 2D grid.

Figure 7c shows forward results of a 2D polygon with horizontally dependent density function; it has been studied by Martin-Atienza and Garcia-Abdeslem (1999), Zhou (2010) and D'Urso (2015), by assuming a density function

$$\rho(\tilde{x}) = 0.5 + 0.02\tilde{x} - 0.02\tilde{x}^2.$$
(52)

Space-domain results are extracted from table 4 in D'Urso (2015). The Conv-Gauss-FFT algorithm with I = K = 2 is applied to calculate gravity anomalies on the 2D grid: x =



Fig. 6 Comparison of space-domain and Conv-Gauss-FFT forward results of the gravity vector component ϕ_z and the gravity gradient tensor component ϕ_{zz} on five profiles *A*, *B*, *C*, *D* and *E* caused by the linear variable density model in Fig. 5. **a** Space-domain forward results of ϕ_z , **b** rms errors of ϕ_z calculated by the Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes, **c** the corresponding errors of the I = K = 10 Conv-Gauss-FFT algorithm, **d** space-domain forward results of ϕ_{zz} , **e** rms errors of ϕ_{zz} calculated by the Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes, **f** the corresponding errors of the I = K = 10 Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes, **f** the corresponding errors of the I = K = 10 Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes, **f** the corresponding errors of the I = K = 10 Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes, **f** the corresponding errors of the I = K = 10 Conv-Gauss-FFT algorithm with increasing number of Gaussian nodes, **f** the corresponding errors of the I = K = 10 Conv-Gauss-FFT algorithm

-5.5:0.01:5.5 km, z = 0:0.1:3 km, after which field values on the profile z = 0 km with horizontal positions coincide with those in D'Urso (2015) are extracted for comparison. The Conv-Gauss-FFT algorithm costs about 0.3 s evaluating the whole 2D grid.



Fig. 7 2D variable density models previously studied by other authors. **a** A 2D prism with quadratic depth-dependent density, **b** a 2D polygon with quadratic depth-dependent density, **c** a 2D polygon with quadratic horizontally dependent density, **d** source body bounded laterally by functions of depth with a second-order polynomial density function

Figure 7d shows forward results of a 2D model bounded laterally by functions of depth. It can be defined in the form of Eq. 41, with $\tilde{z}_1 = 0$ km, $\tilde{z}_2 = 3$ km the upper and bottom depths, and $h_1(\tilde{z}) = -4 - 0.07\tilde{z} + 0.3\tilde{z}^2 + 0.01\tilde{z}^3$ km, $h_2(\tilde{z}) = 4.5 + 0.5\tilde{z} - 0.2\tilde{z}^2$ km the left and right boundary functions; it has been studied by Martin-Atienza and Garcia-Abdeslem (1999), Zhou (2009b, 2010) and D'Urso (2015), by assuming a 2D variable density function

$$\rho(\tilde{x}, \tilde{z}) = -0.7 - 0.05\tilde{x}\tilde{z} + 0.04\tilde{x}^2 + 0.06\tilde{z}^2.$$
(53)

Space-domain results are extracted from table 5 in D'Urso (2015). The Conv-Gauss-FFT algorithm with I = 2, K = 4 is applied to calculate gravity anomalies on the 2D grid: x = -5.5:0.01:5.5 km, z = 0:0.1:3 km, after which field values on the profile z = 0 km

with horizontal positions coincide with those in D'Urso (2015) are extracted for comparison. The Conv-Gauss-FFT algorithm costs about 0.25 s evaluating this model.

In all density functions, the coordinates \tilde{x}, \tilde{z} are expressed in kilometers and the density expressed in g cm⁻³. Figure 7 shows a perfect agreement between the analytical solutions (red and blue lines) and those computed by means of the proposed Conv-Gauss-FFT approach (black dots).

6.2 3D Examples

Now we test the 3D Conv-Gauss-FFT algorithm using 3D models. We choose the triangulated shape model of the asteroid 433 EROS (approximate size $13 \times 13 \times 33$ km), which was mapped by the NEAR Shoemaker probe from April to October 2000 and has been investigated in depth by Zuber et al. (2000), Tsoulis et al. (2009), D'Urso (2014a, b) and Fukushima (2017). Several models of this body at various resolutions have been made available by the Planetary Science Institute (USA) at the address http://www.psi.edu/pds/archive/shape.html. Data can be downloaded at the address http://sbn.psi.edu/pds/asteroid/NEAR_A_5_COLLECTED_MODELS_V1_0/data/msi/ for all resolutions. We compare our results based on the 3D Conv-Gauss-FFT algorithm with space-domain solutions provided in D'Urso (2014a, b) using both constant and variable density functions for the asteroid 433 EROS.

Figure 8 shows forward results of the gravity potential ϕ (units in m² s⁻²) and three gravity vector components ϕ_x , ϕ_y , ϕ_z (units in mGal) on a surface 0.2 km above the surface



Fig. 8 Forward results of the gravity potential ϕ (units in m² s⁻²) and three gravity vector components ϕ_{x} , ϕ_{z} (units in mGal) on a surface 0.2 km above the surface of the asteroid 433 EROS using the 3D Conv-Gauss-FFT algorithm with I = J = K = 2

of the asteroid 433 EROS using the 3D Conv-Gauss-FFT algorithm with I = J = K = 2. Here the asteroid 433 EROS is characterized by 1708 faces and 856 vertices. The results are obtained by assuming $G = 6.67259 \times 10^{-11} \text{ m}^3 \text{ kg}^{-1} \text{ s}^{-2}$ and a constant density $\rho = 2670 \text{ kg m}^{-3}$ of the model.

The computation surface contains 856 vertices, with each vertex 0.2 km above a corresponding vertex of the asteroid 433 EROS. The local normal vector for each vertex is obtained by taking the average direction of the normal vectors of the triangular surfaces sharing this vertex. That is, for each vertex *P* of the asteroid 433 EROS, suppose *P* is shared by *N* surrounding triangular surfaces whose unit normal vectors are \mathbf{n}_i , i = 1, 2, ..., N, then the local unit normal vector corresponding to the vertex *P* is calculated as: $\mathbf{n}_P = \sum_{i=1}^{N} \mathbf{n}_i / |\sum_{i=1}^{N} \mathbf{n}_i|$, and a computation point *Q* with $\mathbf{r}_Q = \mathbf{r}_P + 0.2\mathbf{n}_P$ is evaluated, where \mathbf{r}_P and \mathbf{r}_Q are position vectors for the vertices *P* and *Q*, respectively.

Figure 9 shows the corresponding difference maps between space-domain analytical solutions and the 3D Conv-Gauss-FFT algorithm. For space-domain analytical solution, we have coded the algorithm introduced in D'Urso (2014a) for the gravity potential and its first-order derivatives in MATLAB language and evaluated these components on the computation surface directly. For the 3D Conv-Gauss-FFT algorithm, we first calculate the gravity potential and its first derivatives on the 3D regular grid: x = -20:0.25:20 km, y = -10:0.25:10 km, z = -10:0.25:10 km, and then, field values on the computation surface are obtained through a linear interpolation. The space-domain algorithm costs about 97 s evaluating 856 points. The Conv-Gauss-FFT algorithm costs about 22 s



Fig. 9 Difference maps of the gravity potential ϕ (units in m² s⁻²) and three gravity vector components ϕ_x , ϕ_y , ϕ_z between space-domain analytical solutions and the 3D Conv-Gauss-FFT algorithm with I = J = K = 2

calculating the whole 3D grid, which is of size $81 \times 161 \times 81$ and in all contains 1,056,321 points.

It can be observed that the errors for the gravity potential ϕ and three gravity vector components ϕ_x , ϕ_y , ϕ_z are all very small comparing to the referenced fields. The gravity potential exhibits smaller errors than the gravity vector components because the corresponding integral kernel is smoother and can be better approximated by low-order polynomials. The largest misfits between two algorithms for all three gravity vector components are only about 2 mGal, which proves the accuracy and efficiency of the Conv-Gauss-FFT algorithm, especially when fields on a regular grid are needed.

Some further comparisons between the Conv-Gauss-FFT algorithm and space-domain analytical solutions for the asteroid 433 EROS with constant and variable density contrasts are implemented and summarized in Tables 1 and 2. In all cases, the comparison is made at the origin of the coordinate system, i.e., the point (0, 0, 0). The relative error is used to test the accuracy of the algorithm. Time costs of both algorithms, with T_{space} represents the computation time of the space-domain solution calculating the gravity anomalies at a single point, i.e., the origin, and T_{CGF} represents the computation time of the Conv-Gauss-FFT algorithm evaluating the whole 3D grid mentioned above, are provided.

Table 1 summarizes forward results of both algorithms for the gravity potential ϕ and three gravity vector components ϕ_x , ϕ_y , ϕ_z caused by the asteroid 433 EROS with constant density $\rho = 2670$ kg m⁻³ and characterized by 1708, 7790, 10,152, 22,540, 89,398 and 200,700 faces. For the gravity potential, relative errors for all six models are below 10⁻⁴; for vector gravity components, the largest relative error among all six models is about 0.4%. The computation time for the space-domain solution increases significantly when the number of faces of the polyhedral model becomes larger, while the computation time of the Conv-Gauss-FFT algorithm changes little. This is because in the Conv-Gauss-FFT algorithm, only the point-in-polyhedron algorithm depends on the geometry of the source, and it runs very fast for test points on a regular grid, with the computation time increases very slowly when the polyhedron is characterized by larger numbers of faces.

Table 2 summarizes forward results of both algorithms caused by the asteroid 433 EROS with linear variable density $\rho = 2670 + \tilde{x} + \tilde{y} + \tilde{z}$ kg m⁻³ and characterized by 1708, 7790, 10,152 and 22,540 faces. The model has been studied in D'Urso (2014b) using several linear density functions. Forward results in table 2 in D'Urso (2014b) are extracted for comparison. It should be noted that since the origin of the coordinate system lies inside the source body, the integral in Eq. 43 for three diagonal gravity gradient tensor components ϕ_{xx} , ϕ_{yy} , ϕ_{zz} suffers from singularity problems which cannot be properly solved by regular quadrature methods, including the rectangle-Gaussian quadrature rule embedded in the Conv-Gauss-FFT algorithm, we provide only forward results of the gravity potential ϕ , three gravity vector components ϕ_x , ϕ_y , ϕ_z , and three off-diagonal gravity gradient tensor components ϕ_{xy} , ϕ_{xz} , ϕ_{yz} here. For the gravity potential, relative errors for all models are below 10^{-4} ; for vector gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 0.4%; for tensor gravity components, the largest relative error is about 2%. Time costs of the Conv-Gauss-FFT algorithm for a

Table 1 Comparison of values of the gravity potential ϕ and three gravity vector components ϕ_x , ϕ_y , ϕ_z at the origin caused by the asteroid 433 Eros with constant density and different numbers of faces (N_F) calculated by the Conv-Gauss-FFT algorithm (I = J = K = 2) and the space-domain algorithm in D'Urso (2014a)

$N_{\rm F}$	Constant density contrast: $\rho = 2670 \mathrm{kg} \mathrm{m}^{-3}$							
	Gravity	D'Urso (2014a)	Conv-Gauss-FFT	Relative error	Time (s)	AU		
$N_{\rm F} = 1708$	ϕ	6.868879e+01	6.869223e+01	5.02e-05	$T_{\text{space}} = 0.1$	$m^2 s^{-2}$		
	ϕ_x	1.768255e-04	1.768960e-04	3.99e-04		${\rm ms^{-2}}$		
	ϕ_y	7.775899e-04	7.770288e-04	7.22e-04	$T_{\rm CGF} = 22.1$	${\rm ms^{-2}}$		
	ϕ_z	-1.353872e-04	-1.354283e-04	3.03e-04		${\rm ms^{-2}}$		
$N_{\rm F} = 7790$	ϕ	6.928423e+01	6.927802e+01	8.96e-05	$T_{\rm space} = 0.5$	${\rm m}^2{\rm s}^{-2}$		
	ϕ_x	1.758138e-04	1.752970e-04	2.94e-03		${\rm ms^{-2}}$		
	ϕ_y	7.780774e-04	7.774987e-04	7.44e-04	$T_{\rm CGF} = 26.1$	${\rm ms^{-2}}$		
	ϕ_z	-1.384690e-04	- 1.383389e-04	9.40e-04		${\rm ms^{-2}}$		
$N_{\rm F} = 10,152$	ϕ	6.932709e+01	6.932796e+01	1.25e-05	$T_{\rm space} = 0.7$	$m^2 s^{-2}$		
	ϕ_x	1.755377e-04	1.748158e-04	4.11e-03		${\rm ms^{-2}}$		
	ϕ_y	7.787153e-04	7.783384e-04	4.84e-04	$T_{\rm CGF} = 26.2$	${\rm ms^{-2}}$		
	ϕ_z	-1.392375e-04	- 1.391297e-04	7.74e-04		${\rm ms^{-2}}$		
$N_{\rm F} = 22,540$	ϕ	6.939819e+01	6.940094e+01	3.96e-05	$T_{\rm space} = 1.5$	${\rm m}^2{\rm s}^{-2}$		
	ϕ_x	1.752634e-04	1.746698e-04	3.39e-03		${\rm ms^{-2}}$		
	ϕ_y	7.789840e-04	7.791308e-04	1.88e - 04	$T_{\rm CGF} = 27.0$	${\rm ms^{-2}}$		
	ϕ_z	-1.395084e-04	- 1.394981e-04	7.33e-05		${\rm ms^{-2}}$		
$N_{\rm F} = 89,398$	ϕ	6.944107e+01	6.944270e+01	2.35e-05	$T_{\rm space} = 5.9$	$m^2 s^{-2}$		
	ϕ_x	1.750527e-04	1.745571e-04	2.83e-03		${\rm ms^{-2}}$		
	ϕ_y	7.793210e-04	7.793855e-04	8.28e-05	$T_{\rm CGF} = 30.4$	${\rm ms^{-2}}$		
	ϕ_z	-1.397135e-04	-1.398525e-04	9.95e-04		${\rm ms^{-2}}$		
$N_{\rm F} = 200,700$	ϕ	6.944893e+01	6.945076e+01	2.62e-05	$T_{\text{space}} = 13.4$	$m^2 s^{-2}$		
	ϕ_x	1.750299e-04	1.744399e-04	3.37e-03		${\rm ms^{-2}}$		
	ϕ_y	7.794022e-04	7.794026e-04	4.61e-07	$T_{\rm CGF} = 36.7$	${\rm ms^{-2}}$		
	ϕ_z	-1.397703e-04	-1.400328e-04	1.88e-03		${\rm m}{\rm s}^{-2}$		

 T_{space} represents the computation time of the space-domain solution calculating the gravity anomalies at a single point, i.e., the origin, and T_{CGF} represents the computation time of the Conv-Gauss-FFT algorithm evaluating the whole 3D grid, i.e., 1,056,321 points

AU amplitude unit

7 Discussion

Numerical examples have shown that in general, the Conv-Gauss-FFT algorithm achieves high accuracy. However, it may introduce some observable errors under certain conditions. For example, when the computation point lies near the boundary of the source, and when the geometry of the source does not fit perfectly with a rectangular mesh. A possible way to solve this problem may be the combination of the Conv-Gauss-FFT algorithm with analytical solutions or some properly designed quadrature methods to evaluate the long-range

N _F	Variable density contrast: $\rho = 2670 + \tilde{x} + \tilde{y} + \tilde{z} \text{ kg m}^{-3}$							
	Gravity	D'Urso (2014b)	Conv-Gauss-FFT	Relative error	Time (s)	AU		
$N_{\rm F} = 1708$	ϕ	6.869669e+01	6.870013e+01	5.01e-05	$T_{\rm CGF} = 32.4$	$m^2 s^{-2}$		
	ϕ_x	1.890322e-04	1.891038e-04	3.79e-04		${\rm ms^{-2}}$		
	ϕ_y	7.819466e-04	7.813871e-04	7.16e-04		${\rm ms^{-2}}$		
	ϕ_z	-1.294272e-04	-1.294674e-04	3.11e-04		${\rm ms^{-2}}$		
	ϕ_{xy}	-1.993356e-07	-1.994861e-07	7.55e-04		s^{-2}		
	ϕ_{xz}	-7.989745e-09	-7.852553e-09	1.72e-02		s^{-2}		
	ϕ_{yz}	2.589006e-08	2.584185e-08	1.86e-03		s^{-2}		
$N_{\rm F} = 7790$	ϕ	6.929211e+01	6.928587e+01	9.01e-05	$T_{\rm CGF} = 37.4$	$m^2 s^{-2}$		
	ϕ_x	1.881311e-04	1.876148e-04	2.74e-03		${\rm ms^{-2}}$		
	ϕ_y	7.824665e-04	7.818877e-04	7.40e-04		${\rm ms^{-2}}$		
	ϕ_z	-1.324542e-04	- 1.323271e-04	9.59e-04		${\rm ms^{-2}}$		
	ϕ_{xy}	-1.997333e-07	- 1.999730e-07	1.20e-03		s^{-2}		
	ϕ_{xz}	- 8.117869e-09	-8.051218e-09	8.21e-03		s^{-2}		
	ϕ_{yz}	2.741592e-08	2.693504e-08	1.75e-02		s^{-2}		
$N_{\rm F} = 10,152$	ϕ	6.933499e+01	6.933580e+01	1.17e-05	$T_{\rm CGF} = 37.5$	$m^2 s^{-2}$		
	ϕ_x	1.878646e - 04	1.871438e-04	3.84e-03		${\rm m}{\rm s}^{-2}$		
	ϕ_y	7.831055e-04	7.827309e-04	4.78e-04		${\rm ms^{-2}}$		
	ϕ_z	-1.332192e-04	- 1.331118e-04	8.06e-04		${\rm ms^{-2}}$		
	ϕ_{xy}	-1.996421e-07	- 1.998759e-07	1.17e-03		s^{-2}		
	ϕ_{xz}	-8.132773e-09	- 8.045723e $-$ 09	1.07e-02		s^{-2}		
	ϕ_{yz}	2.761244e-08	2.749387e-08	4.29e-03		s^{-2}		
$N_{\rm F} = 22,540$	ϕ	6.940608e+01	6.940880e+01	3.92e-05	$T_{\rm CGF} = 38.5$	$m^2 s^{-2}$		
	ϕ_x	1.876020e - 04	1.870107e-04	3.15e-03		${\rm m}{\rm s}^{-2}$		
	ϕ_y	7.833815e-04	7.835288e-04	1.88e - 04		${\rm ms^{-2}}$		
	ϕ_z	-1.334826e-04	-1.334726e-04	7.48e-05		${\rm ms^{-2}}$		
	ϕ_{xy}	-1.996434e-07	-1.998672e-07	1.12e-03		s^{-2}		
	ϕ_{xz}	-8.161400e-09	- 8.037579e-09	1.52e-02		s^{-2}		
	ϕ_{yz}	2.790263e-08	2.792218e-08	7.01e-04		s^{-2}		

Table 2 Comparison of values of the gravity potential ϕ , three gravity vector components ϕ_x , ϕ_y , ϕ_z , and three off-diagonal gravity gradient tensor components ϕ_{xy} , ϕ_{yz} , ϕ_{yz} at the origin caused by the asteroid 433 Eros with variable density $\rho = 2670 + \tilde{x} + \tilde{y} + \tilde{z}$ kg m⁻³ and different numbers of faces (*N*_F) calculated by the Conv-Gauss-FFT algorithm (*I* = *J* = *K* = 2) and the space-domain results from D'Urso (2014b)

 T_{CGF} represents the computation time of the Conv-Gauss-FFT algorithm evaluating the whole 3D grid, i.e., 1,056,321 points

AU amplitude unit

and close-range contributions separately. Similar techniques have been used in Parker (1995) and Tsoulis (1998, 2001) in solving terrain correction problems where the combination of analytical solutions for an inner zone surrounding the computation point with modified FFT for the computation of the effect of the rest of the DTM are discussed.

Take the general 2D forward problem for example, let \mathscr{P} be the set of all field points, then for a field point $P(x_P, z_P) \in \mathscr{P}$, denote the rectangular region close to *P* by

$$\Omega(P) = \left\{ (x, z) \middle| |x - x_P| \le L_c \Delta x, |z - z_P| \le N_c \Delta z \right\},\tag{54}$$

where L_c, N_c specify the distances along two dimensions within which a close range is defined. As shown in Fig. 1 by setting $L_c = N_c = 1$, field points can be classified into three categories based on three different relations between the source body region Ω_{polygon} and the region $\Omega(P)$:

$$P \in \begin{cases} \mathscr{P}_{\text{in}}, & \text{if } \Omega(P) \cap \Omega_{\text{polygon}} = \Omega(P), \\ \mathscr{P}_{\text{out}}, & \text{if } \Omega(P) \cap \Omega_{\text{polygon}} = \emptyset, \\ \mathscr{P}_{\text{bnd}}, & \text{otherwise.} \end{cases}$$
(55)

Then we can evaluate the long-range contribution by applying the Conv-Gauss-FFT algorithm with a slightly modified kernel $\hat{\mathbf{t}}_{\tilde{\alpha}_{i}\tilde{\gamma}_{i}}^{\text{long}}$:

$$\hat{\mathbf{t}}_{\tilde{\alpha}_{i}\tilde{\gamma}_{k}}^{\text{long}} = \begin{cases} 0, & \text{if } |x_{l} - \tilde{x}_{\tilde{l} + \tilde{\alpha}_{i}}| \leq L_{c}\Delta x \text{ and } |z_{n} - \tilde{z}_{\tilde{n} + \tilde{\gamma}_{k}}| \leq N_{c}\Delta z, \\ \hat{\mathbf{t}}_{\tilde{\alpha}_{i}\tilde{\gamma}_{k}}, & \text{otherwise,} \end{cases}$$
(56)

and evaluate the close-range contribution using either analytical solutions or some properly designed quadrature methods. For example, if $P \in \mathscr{P}_{out}$, clearly the close-range contribution is zero; if $P \in \mathscr{P}_{in}$, close-range contribution can be accurately evaluated using analytical solution of a 2D prism; if $P \in \mathscr{P}_{bnd}$, analytical solution or some properly designed quadrature methods can be applied once the region $\Omega(P) \cap \Omega_{polygon}$ is rigorously defined.

The 3D case can be understood analogously. However, in order to use analytical solutions, an efficient algorithm for the calculation of the geometric intersection of a rectangular prism and a polyhedron is needed, which may cause some extra computational efforts. Besides, analytical solutions can be used only when they are available for the density model under consideration, and they also require considerable computational costs. An optimal trade-off between numerical accuracy and computational efficiency may be obtained by choosing optimal values of the parameters L_c, M_c, N_c .

8 Conclusions

We have reviewed briefly space-domain and Fourier-domain gravity forward algorithms in Cartesian coordinate system, based on which we introduced a Conv-Gauss-FFT algorithm for the accurate and efficient evaluation of continuous convolution-type integrals, and applied it to solve general 2D and 3D gravity forward problems.

The contribution of the Conv-Gauss-FFT algorithm can be summarized as follows:

- 1. The algorithm can be applied to source bodies with arbitrary geometry, with their boundaries represented either by discrete or continuous functions. Therefore, the algorithm can be applied to almost any 2D or 3D model, including single-source bodies like sphere, ellipsoid, cylinder, prism and general polyhedron, combined models like a 2D layer, a 3D terrain, or a general 3D model.
- Although no numerical example of a prism-staked model is provided here, we mention that in fact the algorithm achieves best numerical accuracy and efficiency for this kind of models. On the one hand, since the source is a 3D mesh of prisms, a prefect match

of the rectangle-Gaussian grid and the source's boundaries can be obtained. On the other hand, the point-in-polyhedron algorithm is no longer needed, which can further accelerate the algorithm.

- 3. The algorithm can be applied to arbitrary density distribution that are defined either by discrete or by continuous functions, including polynomial, exponential, hyperbolic, parabolic, or any other analytically defined function. For the 3D prism-stacked model that is widely applied in both forward and inverse problems, since *IJK* sample points are picked within each prism element when an $I \times J \times K$ nodes 3D Conv-Gauss-FFT algorithm is applied, the density function can vary not only from prism to prism, but also within each prism element. This adds more flexibility to the 3D prism-stacked model.
- 4. Theoretically, the algorithm can be extended to any higher-order derivatives of the gravitational potential simply by replacing the kernel function with the corresponding derivative component of Newton's integral kernel. For the gravity potential and its first-order derivatives, the algorithm can obtain high-precision forward results in the whole 2D or 3D space. For gravity gradient tensor components, the algorithm can provide reliable forward results outside the source body, but may fail or need special treatment inside the source body due to singularity problems. For gravity curvature components and higher-order derivatives, we have not done any numerical example, a reasonable guess is that the algorithm works well for field points outside the source body, and it may fail for those lying within.
- 5. Efficiency of the algorithm is guaranteed by the fast computations of the kernel functions of a mass point, a fast point-in-polyhedron algorithm, and the FFT algorithm. For 3D prism-stacked models, the algorithm presented here is more efficient than the mass-prism-based algorithm in Sanso and Sideris (2013) by avoiding the numerous, time-consuming evaluations of the logarithmic and arctangent terms. Besides, the algorithm presented here is more accurate than the algorithm in Tontini et al. (2009) since edge effects now can be completely eliminated. It is also more accurate than the algorithm in Zhang et al. (2015) because instead of a pure rectangle mesh of mass points, we use a hybrid rectangle-Gaussian grid of mass points.

Future works may be dedicated to the application of appropriate numerical quadrature methods for the evaluation of close-range contribution. Further applications of the algorithm for other convolution-type integrals in geophysics and geodesy, like the modeling of magnetic fields, upward and downward continuation of potential fields, and the evaluation of many geodetic integrals under planar assumption, may also be studied.

Acknowledgements The authors are very grateful to two anonymous reviewers for their critique, helpful comments, and valuable suggestions which improved the manuscript significantly. This study was funded by the National Natural Science Foundation of China under Grant No. 41504089.

References

- Asgharzadeh MF, von Frese RRB, Kim HR, Leftwich TE, Kim JW (2007) Spherical prism gravity effects by Gauss–Legendre quadrature integration. Geophys J Int 169(1):1–11. https://doi.org/10.1111/j.1365-246X.2007.03214.x
- Barnett CT (1976) Theoretical modeling of magnetic and gravitational-fields of an arbitrarily shaped 3-dimensional body. Geophysics 41(6):1353–1364. https://doi.org/10.1190/1.1440685
- Bhattacharyya B (1966) Continuous spectrum of the total-magnetic-field anomaly due to a rectangular prismatic body. Geophysics 31(1):97–121

- Bhattacharyya B, Navolio M (1975) Digital convolution for computing gravity and magnetic anomalies due to arbitrary bodies. Geophysics 40(6):981–992
- Blakely RJ (1996) Potential theory in gravity and magnetic applications. Cambridge University Press, Cambridge
- Cai YG, Wang CY (2005) Fast finite-element calculation of gravity anomaly in complex geological regions. Geophys J Int 162(3):696–708. https://doi.org/10.1111/j.1365-246X.2005.02711.x
- Cai HZ, Zhdanov M (2015) Application of Cauchy-type integrals in developing effective methods for depthto-basement inversion of gravity and gravity gradiometry data. Geophysics 80(2):G81–G94. https:// doi.org/10.1190/GEO2014-0332.1
- Casenave F, Métivier L, Pajot-Métivier G, Panet I (2016) Fast computation of general forward gravitation problems. J Geod 90(7):655–675. https://doi.org/10.1007/s00190-016-0900-2
- Chai Y, Hinze WJ (1988) Gravity inversion of an interface above which the density contrast varies exponentially with depth. Geophysics 53(6):837–845
- Chai Y, Jia J (1990) Parker's formulas in different forms and their applications to oil gravity survey. Oil Geophys Prospect 25(3):321–332
- Chakravarthi V, Raghuram HM, Singh SB (2002) 3-D forward gravity modeling of basement interfaces above which the density contrast varies continuously with depth. Comput Geosci 28(1):53–57. https:// doi.org/10.1016/S0098-3004(01)00080-2
- Chappell A, Kusznir N (2008) An algorithm to calculate the gravity anomaly of sedimentary basins with exponential density-depth relationships. Geophys Prospect 56(2):249–258. https://doi.org/10.1111/j. 1365-2478.2007.00674.x
- Chenot D, Debeglia N (1990) 3-dimensional gravity or magnetic constrained depth inversion with lateral and vertical variation of contrast. Geophysics 55(3):327–335. https://doi.org/10.1190/1.1442840
- Conway JT (2015) Analytical solution from vector potentials for the gravitational field of a general polyhedron. Celest Mech Dyn Astron 121(1):17–38. https://doi.org/10.1007/s10569-014-9588-x
- Cooley JW, Tukey JW (1965) An algorithm for the machine calculation of complex Fourier series. Math Comput 19(90):297–301
- Cordell L (1973) Gravity analysis using an exponential density-depth function—San-Jacinto-Graben, California. Geophysics 38(4):684–690. https://doi.org/10.1190/1.1440367
- Deng XL, Grombein T, Shen WB, Heck B, Seitz K (2016) Corrections to "a comparison of the tesseroid, prism and point-mass approaches for mass reductions in gravity field modelling" (Heck and Seitz, 2007) and "optimized formulas for the gravitational field of a tesseroid" (Grombein et al., 2013). J Geod 90(6):585–587. https://doi.org/10.1007/s00190-016-0907-8
- Driscoll JR, Healy DM (1994) Computing Fourier transforms and convolutions on the 2-sphere. Adv Appl Math 15(2):202–250
- D'Urso MG (2013) On the evaluation of the gravity effects of polyhedral bodies and a consistent treatment of related singularities. J Geod 87(3):239–252. https://doi.org/10.1007/s00190-012-0592-1
- D'Urso MG (2014a) Analytical computation of gravity effects for polyhedral bodies. J Geod 88(1):13–29. https://doi.org/10.1007/s00190-013-0664-x
- D'Urso MG (2014b) Gravity effects of polyhedral bodies with linearly varying density. Celest Mech Dyn Astron 120(4):349–372. https://doi.org/10.1007/s10569-014-9578-z
- D'Urso MG (2015) The gravity anomaly of a 2D polygonal body having density contrast given by polynomial functions. Surv Geophys 36(3):391–425. https://doi.org/10.1007/s10712-015-9317-3
- D'Urso MG (2016) A remark on the computation of the gravitational potential of masses with linearly varying density. In: VIII Hotine-Marussi symposium on mathematical geodesy, vol 142, pp 205–212. https://doi.org/10.1007/1345_2015_138
- D'Urso MG, Trotta S (2015) Comparative assessment of linear and bilinear prism-based strategies for terrain correction computations. J Geod 89(3):199–215. https://doi.org/10.1007/s00190-014-0770-4
- D'Urso MG, Trotta S (2017) Gravity anomaly of polyhedral bodies having a polynomial density contrast. Surv Geophys 38(4):781–832. https://doi.org/10.1007/s10712-017-9411-9
- Farquharson C, Mosher C (2009) Three-dimensional modelling of gravity data using finite differences. J Appl Geophys 68(3):417–422. http://www.sciencedirect.com/science/article/pii/ S0926985109000500
- Forsberg R (1985) Gravity field terrain effect computations by FFT. Bull Geod 59(4):342-360
- Fukushima T (2017) Precise and fast computation of the gravitational field of a general finite body and its application to the gravitational study of asteroid eros. Astron J 154(4):145. https://doi.org/10.3847/ 1538-3881/aa88b8
- Gallardo LA, Perez-Flores MA, Gomez-Trevino E (2005) Refinement of three-dimensional multilayer models of basins and crustal environments by inversion of gravity and magnetic data. Tectonophysics 397(1–2):37–54. https://doi.org/10.1016/j.tecto.2004.10.010

- Gallardo-Delgado LA, Perez-Flores MA, Gomez-Trevino E (2003) A versatile algorithm for joint 3D inversion of gravity and magnetic data. Geophysics 68(3):949–959. https://doi.org/10.1190/1.1581067
- Garcia-Abdeslem J (1992) Gravitational attraction of a rectangular prism with depth-dependent density. Geophysics 57(3):470–473
- Garcia-Abdeslem J (2003) 2D modeling and inversion of gravity data using density contrast varying with depth and source-basement geometry described by the Fourier series. Geophysics 68(6):1909–1916. https://doi.org/10.1190/1.1635044
- Garcia-Abdeslem J (2005) The gravitational attraction of a right rectangular prism with density varying with depth following a cubic polynomial. Geophysics 70(6):J39–J42. https://doi.org/10.1190/1.2122413
- Garcia-Abdeslem J, Romo JM, Gomez-Trevino E, Ramirez-Hernandez J, Esparza-Hernandez FJ, Flores-Luna CF (2005) A constrained 2D gravity model of the Sebastián Vizcaíno Basin, Baja California Sur, Mexico. Geophys Prospect 53(6):755–765
- Granser H (1987) 3-dimensional interpretation of gravity-data from sedimentary basins using an exponential density depth function. Geophys Prospect 35(9):1030–1041. https://doi.org/10.1111/j.1365-2478.1987. tb00858.x
- Grombein T, Seitz K, Heck B (2013) Optimized formulas for the gravitational field of a tesseroid. J Geod 87(7):645–660. https://doi.org/10.1007/s00190-013-0636-1
- Gruber C, Novák P, Sebera J (2011) FFT-based high-performance spherical harmonic transformation. Studia Geophys Geod 55(3):489–500
- Guptasarma D, Singh B (1999) New scheme for computing the magnetic field resulting from a uniformly magnetized arbitrary polyhedron. Geophysics 64(1):70–74. https://doi.org/10.1190/1.1444531
- Guspi F (1992) 3-dimensional Fourier gravity inversion with arbitrary density contrast. Geophysics 57(1):131–135
- Hamayun Prutkin I, Tenzer R (2009) The optimum expression for the gravitational potential of polyhedral bodies having a linearly varying density distribution. J Geod 83(12):1163–1170. https://doi.org/10. 1007/s00190-009-0334-1
- Hansen RO (1999) An analytical expression for the gravity field of a polyhedral body with linearly varying density. Geophysics 64(1):75–77. https://doi.org/10.1190/1.1444532
- Hansen R, Wang X (1988) Simplified frequency-domain expressions for potential fields of arbitrary threedimensional bodies. Geophysics 53(3):365–374
- Heck B, Seitz K (2007) A comparison of the tesseroid, prism and point-mass approaches for mass reductions in gravity field modelling. J Geod 81(2):121–136. https://doi.org/10.1007/s00190-006-0094-0
- Hirt C, Featherstone WE, Claessens SJ (2011) On the accurate numerical evaluation of geodetic convolution integrals. J Geod 85(8):519–538. https://doi.org/10.1007/s00190-011-0451-5
- Holstein H (2003) Gravimagnetic anomaly formulas for polyhedra of spatially linear media. Geophysics 68(1):157–167. https://doi.org/10.1190/1.1543203
- Holstein H, Ketteridge B (1996) Gravimetric analysis of uniform polyhedra. Geophysics 61(2):357–364. https://doi.org/10.1190/1.1443964
- Jia Z, Wu SG (2011) Potential fields and their partial derivatives produced by a 2D homogeneous polygonal source: a summary with some revisions. Geophysics 76(4):L29–L34. https://doi.org/10.1190/1. 3587221
- Jiang L, Zhang J, Feng Z (2017) A versatile solution for the gravity anomaly of 3D prism-meshed bodies with depth-dependent density contrast. Geophysics 82(4):G77–G86. https://doi.org/10.1190/geo2016-0394.1
- Lee TC, Biehler S (1991) Inversion modeling of gravity with prismatic mass bodies. Geophysics 56(9):1365–1376. https://doi.org/10.1190/1.1443156
- Li X, Chouteau M (1998) Three-dimensional gravity modeling in all space. Surv Geophys 19(4):339–368. https://doi.org/10.1023/A:1006554408567
- Li YC, Sideris MG (1994) Improved gravimetric terrain corrections. Geophys J Int 119(3):740–752. https:// doi.org/10.1111/j.1365-246X.1994.tb04013.x
- Martin-Atienza B, Garcia-Abdeslem J (1999) 2-D gravity modeling with analytically defined geometry and quadratic polynomial density functions. Geophysics 64(6):1730–1734
- Mohlenkamp MJ (1999) A fast transform for spherical harmonics. J Fourier Anal Appl 5(2-3):159-184
- Murthy IR, Rao DB (1979) Gravity anomalies of two-dimensional bodies of irregular cross-section with density contrast varying with depth. Geophysics 44(9):1525–1530
- Nagy D (1966) The gravitational attraction of a right rectangular prism. Geophysics 31(2):362-371
- Nagy D, Papp G, Benedek J (2000) The gravitational potential and its derivatives for the prism. J Geod 74(7–8):552–560
- Okabe M (1979) Analytical expressions for gravity anomalies due to homogeneous polyhedral bodies and translations into magnetic anomalies. Geophysics 44(4):730–741

Parker R (1973) The rapid calculation of potential anomalies. Geophys J R Astron Soc 31(4):447–455

- Parker RL (1995) Improved Fourier terrain correction. 1. Geophysics 60(4):1007–1017. https://doi.org/10. 1190/1.1443829
- Pedersen LB (1978) Wavenumber domain expressions for potential fields from arbitrary 2-, 21/2-, and 3-dimensional bodies. Geophysics 43(3):626–630
- Pohanka V (1998) Optimum expression for computation of the gravity field of a polyhedral body with linearly increasing density. Geophys Prospect 46(4):391–404. https://doi.org/10.1046/j.1365-2478. 1998.960335.x
- Rao DB (1986) Modelling of sedimentary basins from gravity anomalies with variable density contrast. Geophys J R Astron Soc 84(1):207–212
- Rao CV, Pramanik AG, Kymar GVRK, Raju MLML (1994) Gravity interpretation of sedimentary basins with hyperbolic density contrast. Geophys Prospect 42(7):825–839. https://doi.org/10.1111/j.1365-2478.1994.tb00243.x
- Ren Z, Chen C, Pan K, Kalscheuer T, Maurer H, Tang J (2017a) Gravity anomalies of arbitrary 3D polyhedral bodies with horizontal and vertical mass contrasts. Surv Geophys 38(2):479–502. https:// doi.org/10.1007/s10712-016-9395-x
- Ren Z, Tang J, Kalscheuer T, Maurer H (2017b) Fast 3D large-scale gravity and magnetic modeling using unstructured grids and an adaptive multilevel fast multipole method. J Geophys Res 122(1):79–109
- Ren Z, Zhong Y, Chen C, Tang J, Pan K (2017c) Gravity anomalies of arbitrary 3D polyhedral bodies with horizontal and vertical mass contrasts up to cubic order. Geophysics. https://doi.org/10.1190/geo2017-0219.1
- Rokhlin V, Tygert M (2005) Fast algorithms for spherical harmonic expansions. Society for Industrial and Applied Mathematics
- Sanso F, Sideris MG (2013) Geoid determination: theory and methods. Lecture notes in earth system sciences. Springer, Berlin
- Sideris MG, Li YC (1993) Gravity-field convolutions without windowing and edge effects. Bull Geod 67(2):107–118. https://doi.org/10.1007/BF01371374
- Smith DA (2000) The gravitational attraction of any polygonally shaped vertical prism with inclined top and bottom faces. J Geod 74(5):414–420
- Talwani M, Worzel JL, Landisman M (1959) Rapid gravity computations for two-dimensional bodies with application to the Mendocino submarine fracture zone. J Geophys Res 64(1):49–59
- Tontini FC, Cocchi L, Carmisciano C (2009) Rapid 3-D forward model of potential fields with application to the Palinuro Seamount magnetic anomaly (southern Tyrrhenian Sea, Italy). J Geophys Res Solid Earth 114(B02):103. https://doi.org/10.1029/2008JB005907
- Tsoulis DV (1998) A combination method for computing terrain corrections. Phys Chem Earth 23(1):53-58
- Tsoulis D (2001) Terrain correction computations for a densely sampled DTM in the Bavarian Alps. J Geod 75(5–6):291–307. https://doi.org/10.1007/s001900100176
- Tsoulis D (2012) Analytical computation of the full gravity tensor of a homogeneous arbitrarily shaped polyhedral source using line integrals. Geophysics 77(2):F1–F11. https://doi.org/10.1190/GEO2010-0334.1
- Tsoulis D, Petrovic S (2001) On the singularities of the gravity field of a homogeneous polyhedral body. Geophysics 66(2):535–539. https://doi.org/10.1190/1.1444944
- Tsoulis D, Wziontek H, Petrovic S (2003) A bilinear approximation of the surface relief in terrain correction computations. J Geod 77(5–6):338–344. https://doi.org/10.1007/s00190-003-0332-7
- Tsoulis D, Jamet O, Verdun J, Gonindard N (2009) Recursive algorithms for the computation of the potential harmonic coefficients of a constant density polyhedron. J Geod 83(10):925–942. https://doi.org/10.1007/s00190-009-0310-9
- Tygert M (2006) Fast algorithms for spherical harmonic expansions. II. J Comput Phys 227(8):4260-4279
- Uieda L, Barbosa VCF, Braitenberg C (2016) Tesseroids: forward-modeling gravitational fields in spherical coordinates. Geophysics 81(5):F41–F48
- Vogel CR (2002) Computational methods for inverse problems, vol 23. SIAM, Philadelphia
- Werner RA (2017) The solid angle hidden in polyhedron gravitation formulations. J Geod 91(3):307–328. https://doi.org/10.1007/s00190-016-0964-z
- Wu XZ (1981) Computation of spectrum of potential field due to 3-dimensional bodies (homogeneous models). Chin J Geophys Chin Ed 24(3):336–348
- Wu XZ (1983) The computation of spectrum of potential-field due to 3-D arbitrary bodies with physical parameters varying with depth. Chin J Geophys Chin Ed 26(2):177–187
- Wu L (2016) Efficient modelling of gravity effects due to topographic masses using the Gauss-FFT method. Geophys J Int 205(1):160–178

- Wu L, Chen L (2016) Fourier forward modeling of vector and tensor gravity fields due to prismatic bodies with variable density contrast. Geophysics 81(1):G13–G26
- Wu L, Lin Q (2017) Improved Parker's method for topographic models using Chebyshev series and low rank approximation. Geophys J Int 209(2):1296–1325
- Wu L, Tian G (2014) High-precision Fourier forward modeling of potential fields. Geophysics 79(5):G59–G68. https://doi.org/10.1190/GEO2014-0039.1
- Zhang J, Jiang L (2017) Analytical expressions for the gravitational vector field of a 3-D rectangular prism with density varying as an arbitrary-order polynomial function. Geophys J Int 210(2):1176–1190. https://doi.org/10.1093/gji/ggx230
- Zhang Y, Wong YS (2015) BTTB-based numerical schemes for three-dimensional gravity field inversion. Geophys J Int 203(1):243–256. https://doi.org/10.1093/gji/ggv301
- Zhang JZ, Zhong BS, Zhou XX, Dai Y (2001) Gravity anomalies of 2-D bodies with variable density contrast. Geophysics 66(3):809–813. https://doi.org/10.1190/1.1444970
- Zhang EH, Shi L, Li YH, Wang QS, Han CW (2015) 3D interface inversion of gravity data in the frequency domain using a parabolic density-depth function and the application in Sichuan-Yunnan region. Chin J Geophys Chin Ed 58(2):556–565
- Zhdanov MS, Liu XJ (2013) 3-D Cauchy-type integrals for terrain correction of gravity and gravity gradiometry data. Geophys J Int 194(1):249–268. https://doi.org/10.1093/gji/ggt120
- Zhou XB (2008) 2D vector gravity potential and line integrals for the gravity anomaly caused by a 2D mass of depth-dependent density contrast. Geophysics 73(6):143–I50. https://doi.org/10.1190/1.2976116
- Zhou XB (2009a) 3D vector gravity potential and line integrals for the gravity anomaly of a rectangular prism with 3D variable density contrast. Geophysics 74(6):143–153. https://doi.org/10.1190/1.3239518
- Zhou XB (2009b) General line integrals for gravity anomalies of irregular 2D masses with horizontally and vertically dependent density contrast. Geophysics 74(2):I1–I7. https://doi.org/10.1190/1.3073761
- Zhou X (2010) Analytic solution of the gravity anomaly of irregular 2D masses with density contrast varying as a 2D polynomial function. Geophysics 75(2):111–119. https://doi.org/10.1190/1.3294699
- Zuber MT, Smith DE, Cheng AF, Garvin JB, Aharonson O, Cole TD, Dunn PJ, Guo Y, Lemoine FG, Neumann GA, Rowlands DD, Torrence MH (2000) The shape of 433 Eros from the NEAR-Shoemaker laser rangefinder. Science 289(5487):2097. http://science.sciencemag.org/content/289/5487/2097. abstract